Racial Disparities in the Income Tax Treatment of Marriage*

Janet Holtzblatt

Swati Joshi

Nora Cahill

William Gale

October 18, 2023

*The authors are affiliated with the Urban-Brookings Tax Policy Center (TPC). Holtzblatt is a senior fellow at the Urban Institute. Joshi was a senior research analyst at the Brookings Institution and is now a graduate student at the University of Wisconsin. Cahill was a senior research assistant at Brookings and is now attending the Harvard Kennedy School. Gale is the Miller Chair at Brookings and codirector of the TPC. The authors thank Arnold Ventures and the Robert Wood Johnson Foundation for financial support. This project was part of "Interrupting Structural Racism," a project coordinated by Urban's Office of Race and Equity Research that is demonstrating various approaches to exposing the impacts of structural racism and informing potential remedies. For helpful guidance and comments, the authors thank Chitra Balakrishnan, Rekha Balu, Ian Berlin, Dorothy Brown, Claire Cusella, Elena Derby, Luisa Godinez-Puig, Robert Moffitt, Justyce Watson, and members of the TPC Racial Equity Strategic Planning Advisory Group. The authors extend a special thanks to Dan Feenberg for programming a two-earner deduction in TAXSIM for purposes of this paper.

ABSTRACT

Although it is generally blind with respect to race, the federal individual income tax can create racial disparities when factors that affect tax liability are associated with race. We provide new evidence on racial differences in marriage penalties and bonuses in the income tax, using data from eight waves of the Survey of Consumer Finances. Our results support Brown's (2021) hypothesis that, controlling for income, penalties are more frequent and larger for Black couples than white couples. We link these results to racial differences in relative spousal earnings, the presence of dependents, and the level of income. We show that marriage rates are much higher among white adults than Black adults, which implies that two policy reforms we examine end up benefiting a greater share of white adults than Black adults.

I. INTRODUCTION

The individual income tax generally does not refer to race, and the relationship between a taxpayer's race and their tax bill has largely been ignored by policymakers and researchers (Bearer-Friend, 2019; Gale, 2021). Nevertheless, the income tax can create and reinforce racial disparities due to the way factors that affect taxes are correlated with race.

For example, the tax treatment of marriage has attracted attention for decades, but only recently have scholars—most notably Brown (1997a; 1997b; 1999a; 1999b; 2005; 2021) and Moran and Whitford (1996)—begun to address the racial dimensions of this issue. Marriage penalties, bonuses, or both will exist in any progressive, family-based income tax. Generally, under US tax law, controlling for family income, couples with more equal incomes are more likely to face penalties than those with less equal incomes (throughout the paper, we refer to married couples as "couples"). Brown (2021) and others show that Black women are more likely to work than white women and are paid more, relative to Black men, than white women earn relative to white men. As a result, she hypothesizes that, controlling for family income, Black couples are more likely to face marriage penalties and to face larger marriage penalties than white couples.

We note two other reasons why Black and white couples may face different tax implications of marriage. First, the groups have differences in family characteristics, such as the presence of children, and various features of the tax code affecting children are not neutral with respect to marriage. Second, Black couples and white couples have different distributions of income, and certain tax provisions that are not marriage-neutral are concentrated at specific income levels.

In this paper, we explore how the federal individual income tax treats Black couples

relative to white couples. We use data from eight waves of the Survey of Consumer Finances (SCF), a public-use triennial household survey that contains information on race, demographics, income, and wealth. We split the household data into tax units using a methodology developed in Gale et al. (2022a, b). We calculate marriage penalties and bonuses building off the approach used in Bull et al. (1999) and applying the data to the National Bureau of Economic Research's TAXSIM model.

We obtain several principal findings. First, on an overall basis, Black couples face higher tax costs of marriage. Black couples were more likely to face penalties (46 percent to 43 percent) and less likely to receive bonuses (36 percent to 43 percent) than white couples were, under 2018 law. We compute marriage penalties and bonuses both in dollar terms and as a "rate"—a share of income. We believe the latter measure is more informative both because the "rate" adjusts for the couple's resources and because the distribution of income differs between Black couples and white couples. Among those with penalties, relative to white couples, Black couples paid less in dollars (\$1,804 versus \$2,091) but paid more as a share of income (1.8 percent versus 1.4 percent). Among those with bonuses, the bonus was smaller for Black couples than white couples (\$1,926 versus \$3,304) but the bonus rate was about the same: 2.6 percent for Black couples and 2.7 percent for white couples. All these differences except the last are statistically significant.

Second, our results support Brown's (2021) hypothesis: Controlling for family income, penalties are more prevalent and larger for Black couples than white couples. For example, under 2018 tax law, among tax units with adjusted gross income (AGI) between \$50,000 and \$100,000, and relative to white couples, Black couples were more likely to face marriage penalties (59 percent to 51 percent) and less likely to receive marriage bonuses (33 percent to 44 percent). Among those with a penalty, Black couples faced higher average penalties (\$1,394 versus \$1,241). Among those with a bonus, Black couples received smaller average bonuses (\$1,402 versus \$1,576). Taken together, Black couples in this income group paid, on average, a net penalty of \$358. White couples in this income group received, on average, a net bonus of \$61. All the differences are highly statistically significant. Results from other income groups—except the lowest, where a sizable fraction of couples do not owe any income tax—are generally consistent with these patterns.

Third, we show that, controlling for income, marriage penalties are substantially higher for couples with relatively equal earnings and couples with dependents, and that Black couples are more likely than white couples to have more equal earnings (both because they are more likely to have two earners and because, among two-earner couples, Black couples have more equal earnings than white couples do) and more likely to have dependents.

In regression analysis, we show that the difference in the prevalence of marriage penalties between Black couples and white couples is driven by racial differences in the level of income, the prevalence of two-earner families, and the presence of dependents.

Fourth, we highlight the importance of different marriage rates by race. In the SCF data we use, couples represented 24 percent of Black tax units and 38 percent of Black adults (persons over the age of 18) but 52 percent of white tax units and 68 percent of white adults. As a result, although a greater share of Black couples than white couples experience penalties, a greater share of white tax units and white adults incur marriage penalties than Black tax units and adults. Consequently, policies to reduce marriage penalties, even if they are motivated by racial disparities in marriage penalties, may nevertheless provide tax cuts for a greater share of white tax units.

Fifth, we examine how changes in tax law have affected marriage penalties over time, by comparing penalties under 2018 tax law with the penalties that arise due to the tax laws operating in 2000 and 2015. The 2001 tax cut reduced marriage penalties by changing the two lowest income tax brackets, the standard deduction, and the phaseout rules for the earned income tax credit (EITC). Rate cuts in 2001 and 2017 and other changes in those years, as well as in 2003 and 2009, affected the tax treatment of marriage as well. We find that the evolution of tax changes generally reduced the overall prevalence of marriage penalties. About 56 percent of Black and white couples in our sample faced marriage penalties under 2000 law, compared to 47 percent under 2015 law and 43 percent under 2018 law. But the tax changes raised the gap between Black and white couples. Both Black and white couples incurred, on average, a net penalty of 0.9 percent of income under 2000 law, falling to 0.1 percent under 2015 law. The Tax Cuts and Jobs Act of 2017 (TCJA), however, boosted net marriage bonuses for white couples—and particularly those with high incomes—with no discernable effect, on average, for Black couples.

Sixth, we examine two potential policy changes. Since the main drivers of penalties, other than income, are dependents and relatively equal spousal earnings, we examine reforms that could address each cause.

Giving spouses the option to file individually rather than jointly would help reduce the impact of dependents on marriage penalties. The spouse who claimed the child or children could file as a head of household and possibly qualify for the CTC and/or EITC (or a larger credit amount) based solely on their own income.¹ The policy would eliminate *all* marriage penalties

¹ The Tax Relief and Jobs Cuts Act of 2017 eliminated marriage penalties in the phaseout range of the child tax credit through 2025. The American Rescue Plan of 2021 expanded the child tax credit (for one year) but reintroduced marriage penalties in the phaseout range. President Biden has proposed extending that expansion (and thus the marriage penalties, as well) through the end of 2025."

but would be expensive, with an annual cost of about \$49 billion (in 2018 dollars). Black couples and white couples would receive gains of 1.1 percent and 0.8 percent of income, respectively. The changes would be larger among couples with dependents and larger, as a share of income, among lower-income groups: the policy would be progressive among married couples. But the tax cut would only go to 10 percent of Black tax units (17 percent of Black adults), compared to 22 percent of white tax units (30 percent of white adults).

Reinstating a two-earner deduction, similar to the one in place in the early 1980s, could help reduce the impact of relatively equal spousal earnings on marriage penalties. Such a policy would have reduced total penalties by nearly \$15 billion and increased bonuses by about \$5 billion in 2018. But the prevalence of marriage penalties would have fallen by only 3.1 percentage points among white couples and 1.8 percentage points among Black couples. Even among two-earner couples, the prevalence of penalties would fall by less than 7 percentage points. The effects are quite large, however, among the highest income groups, both because they are more likely to use the deduction and because the deduction is worth more to couples in higher tax brackets. That is, the policy is regressive among married couples. The benefits would go to 13 percent of Black tax units (20 percent of Black adults), compared to 25 percent of white tax units (33 percent of white adults).

Our results are similar in many ways to those in Alm et al. (2023), who used data from the Current Population Survey for a systematic empirical investigation of race and the income tax treatment of marriage. Their work and ours finds similar results for: the relative proportion of Black and white couples with marriage penalties in 2018; the average size of penalties and bonuses in 2018 among those who experience them; declines in the prevalence of marriage penalties due to changes in tax law since 2000; and effects from TCJA of 2017 that helped white

5

couples more than Black couples. Our study and theirs highlight the role of relative spousal earnings in determining marriage penalties, consistent with Brown (2021).

Relative to their work, our paper makes several contributions. First, we provide independent confirmation of their results using a different data set. Second, we highlight the key role that dependents play in the determination of marriage penalties and bonuses. Third, we show in regression analysis that racial differences in the prevalence of marriage penalties can be largely explained by factors that are correlated with race—income levels, spousal earnings shares, and dependents. Fourth, we estimate the effects of two policy reforms and highlight the role that differential marriage rates among Black and white adults play in determining who receives the benefits.

Section II explains how marriage bonuses and penalties can arise in an individual income tax system. Section III describes the data and methodology. Section IV presents the results under 2018 law. Section V presents results under previous years' tax laws and under the two policy reforms mentioned above. Section VI discusses the marriage penalty in the broader context of structural racism and lays out directions for future research.

II. THE INCOME TAX TREATMENT OF MARRIAGE

A couple is said to face a marriage penalty if they owe more individual income tax if they are married than they would if they were single. Conversely, a couple has a marriage bonus if they would owe less income tax when married than they would if single.²

A reasonable starting point is to ask why the tax system favors or disfavors marriage at

² Typically, these comparisons are made assuming that married couples file a joint return and individuals file as single or, if children are present, as head of household. In practice, married couples are permitted to file individually under a "married, filing separately" status, but usually tax liability is smaller under a "married, filing jointly" status. For example, taxpayers who file separately cannot claim the earned income tax credit under current law.

all. The answer is that marriage neutrality—the absence of either marriage penalties or subsidies—runs up against other tax policy goals. Specifically, any tax that applies to family income cannot simultaneously be progressive, provide equal taxation of families with the same income, and remain marriage neutral (Bittker 1975, Rosen 1987). Thus, marriage neutrality conflicts with common notions of vertical and horizontal equity.³

Consider, for example, a progressive tax that imposes the same burden on families with the same income. In this simple example, the first \$20,000 of income is exempt from tax; the next \$20,000 faces a 10 percent rate; and all additional income is subject to a 20 percent rate. The brackets and rates are the same for single and married filers. There are no dependents, exemptions, deductions, or credits (see Table 1).

Under this system, every married couple with the same income has the same tax liability. Marriage penalties arise because *unmarried* couples with the same combined income face different combined tax liability, depending on the allocation of income across partners. Consider the combined taxes of three unmarried couples, each with \$60,000 in income:

- A couple with one earner would owe \$6,000;
- A couple with two earners who earn unequal amounts—one earning \$50,000 and the other earning \$10,000—would owe \$4,000; and
- A couple with two equal earners—each earning \$30,000—would owe \$2,000.

If the couples married, the combined tax owed would not change for the one-earner couple; it would rise by \$2,000 for the couple with unequal earnings and by \$4,000 for the

³ The notion of imposing the same tax burden on families with the same income can be expanded to cover families in similar circumstances, but the definition of what constitutes similarity is vague and controversial. The tax code mostly uses household income, filing status, and number of children to identify families with similar circumstances. Brown (2021) makes the point that two-earner and single-earner couples with the same household financial income should not be seen as similar for horizontal equity purposes, since the non-working spouse in the single-earner couple can more easily provide household services than either working spouse in the two-earner couple.

couple with equal earnings. That is, a family-based progressive tax imposes greater penalties on couples with more equal earnings.

Modifying the rates and brackets could eliminate marriage penalties but could create marriage bonuses. For example, setting the width of the tax brackets for single filers to be half the levels as for married couples would eliminate marriage penalties entirely but would lead to a \$3,000 marriage bonus for the one-earner couple and a \$1,000 marriage bonus for the couple where the partners earn \$50,000 and \$10,000. Other policy changes could generate marriage penalties for some couples and bonuses for others. While creating a bonus for marriage may not appear problematic at first glance, it implies a tax penalty for being single, which helps illustrate the difficult nature of the problem from an equity perspective.

Marriage penalties and bonuses can vary for several reasons besides relative spousal earnings. First, income levels can influence the presence and magnitude of marriage penalties. In the example above, if all incomes were cut in half, the marriage penalties would be cut in half. Second, any deduction or credit whose value is not double for married couples relative to single filers will create penalties. For example, the itemized deduction for state and local property, income, and sales taxes is capped at \$10,000 for both married couples and for single filers, thus creating a marriage penalty. In tax year 2022, the standard deduction was \$25,900 for married couples, half that for single filers, and \$19,400 under the head-of-household filing status, thus creating a marriage penalty for couples with children.⁴ Policies that assist low- and middle-income parents and workers can also create penalties or bonuses.⁵

⁴ By maintaining the home in which their children live, custodial unmarried parents and others caring for dependent relatives can qualify for a larger standard deduction and a more generous tax rate schedule than other unmarried individuals, who must file as singles.

⁵ Bonuses can arise, for example, if a low-income childless worker marries a nonworking parent. Under current law, if single, the worker could be eligible for the small EITC available to very low-income workers who do not live with

Overall, there are three main reasons for differential marriage penalties for Black couples and white couples—differences in income levels, relative spousal earnings, and the presence of dependents. Figure 1 shows how these characteristics differ between Black and white couples in the sample that we use (and explain further below). First, Black couples generally have lower income than white couples. About 37 percent of Black couples have AGI below \$50,000, compared to 28 percent of white couples. About 4 percent of Black couples have AGI above \$200,000, compared to 13 percent of white couples. Second, controlling for income, Black couples are 7-9 percentage points more likely to have two earners. (The overall difference is just 3 percentage points because the distribution of income differs across racial groups.) Third, and again controlling for income, Black couples are 8-16 percentage points more likely to have dependents.

To remove marriage penalties and bonuses completely, policymakers would have to either eliminate the progressivity of the income tax or violate the idea that families with equal income and similar circumstances should pay equal taxes. In practice, the tax system does violate the latter condition, in part because of beliefs about the incentive effects of marriage penalties and the social benefits of marriage. Marriage has been linked to positive personal and social outcomes for couples and their children.⁶ It is unclear, however, whether the association is due to

children, while the nonworking parent would not be eligible for any EITC at all. If married, the couple would receive a substantially larger EITC than the working spouse received on their own as well as the child tax credit. In contrast, the phaseouts of the EITC and (if TCJA expires as scheduled in 2025, the child tax credit) can create marriage penalties, especially for two-earner couples. By filing jointly, couples combine their income, which can push them higher up (or completely out of) the phaseout range than if they filed as if they were not married (Holtzblatt and Rebelein, 2000).

⁶ Marriage penalties reduce the incentive to marry, but empirical studies generally find that penalties have only a small effect on marriage decisions with a somewhat larger impact on the timing of marriage (Alm and Whittington, 1995, 2003; Sjoquist and Walker, 1995). The outcomes in question include higher educational attainment and earnings for children, better physical and mental health for parents and children, and more stable relationships for the children of married couples (see, e.g., Brown, 2010; Howard and Reeves, 2014; Kearney and Levine, 2017; McLanahan and Sawhill, 2015; Ribar, 2015; Wilcox, Lerman, and Price, 2015).

marriage itself or whether those who are likely to succeed are also more likely to marry.⁷ Furthermore, the limited research that distinguishes effects by race shows that the positive associations between marriage and well-being apply mainly to white households.⁸

III. DATA AND METHODOLOGY

No publicly available data set contains information about both race and taxes. To address this shortfall, we proceed in several steps. To generate household-level data on race, other demographic characteristics, income, and wealth, we access data from eight waves (1998 to 2019) of the Survey of Consumer Finances (SCF). To form income tax units out of the SCF households, we use the methodology developed in Gale et al. (2022a, b).⁹ To determine how to treat married couples under the counterfactual that they were unmarried, we generally follow the methodology developed in Bull et al. (1999). To compute federal income tax liabilities. we apply the National Bureau of Economic Research TAXSIM microsimulation model (Feenberg and Coutts, 1993).¹⁰

⁷ Affluent Americans are more likely to get married and stay married, confounding any effect that marriage may inherently have on spouses' and children's wellbeing (Brown, 2010; Howard and Reeves, 2014).

⁸ See, for example, Brown (2010) and Shapiro et al. (2013). In one of the few studies that asked whether Black and white individuals responded differently in response to the marriage disincentives embedded in the tax code, Alm and Whittington (1999) did not find any distinctions between Black and white women but found that the probability of marriage for Black men actually rose by a "quite small" amount as marriage penalties increased.

⁹ An income tax unit is defined as an individual or married couple who is required to file a tax return, or who would be required to file a tax return if their income were high enough, along with all dependents of that individual or married couple.

¹⁰ Our approach—imputing taxes onto a data set that already contains information about race—captures the respondents' self-reported race and, although it may not generate the exact liability a tax unit faces, the error is likely to be small. Other empirical strategies offer different strengths and weaknesses. For example, starting with a data set that has tax information would provide precise more information on taxes but requires imputations for race. With access to confidential tax data, Treasury Department economists have imputed race to the Office of Tax Analysis's microsimulation model by using a set of explanatory variables, including the taxpayer's sex, first and last names, and zip code, to make inferences about the person's race and Hispanic origin and then applying Bayesian inference to estimate the probabilities that each taxpayer falls into a race or ethnic category (Cronin et al, 2023; Fisher, 2023).Using household survey data, the Tax Policy Center is have developed its first race imputations to apply to its microsimulation model (Khitatrakun, Mermin, Page, and Rohaly, 2023). Both organizations are still evaluating and refining their methodologies and have, as of April 2023, limited the type of analysis conducted with

A. Data

To form income tax units out of the SCF households, we proceed in several steps. First, we divide SCF households into tax units as explained in Gale et al. (2022a). For households that account for the vast majority of income—including singles living alone and married couples with either no dependents or with children younger than 18—this process is simple. For other households, a variety of financial and demographic measures are used to estimate filing status.

Second, because the SCF's measures of income and itemizable expenses do not always align with tax concepts and because some variables—including net business income and respondent's age—are intentionally rounded or masked in the public version to avoid reidentification of the participants, we derive or estimate (using other variables in the data set) several items that are needed to compute tax liability.¹¹

Third, after determining how to treat married couples if they were single (as described below), we run the data for tax units through TAXSIM to estimate federal income tax liabilities. TAXSIM follows US federal and state income tax rules over the 1998 to 2019 period (tax years 1997 to 2018) spanned by the SCF data.

In general, we replicate the revenue and distributional effects of the actual system fairly closely. Appendix Figure 1 shows that the difference between estimated revenues in Gale et al. (2022b) (the red line) and the reported revenues from tax return data in the IRS's Statistics of

the new imputations. Alternatively, linking data sets with information on race (e.g., the Census) and taxes (e.g., tax returns) would provide greater accuracies on both measures, but the amount of tax data provided by the IRS to the Census department is limited by law and regulations and further would not provide data on household wealth that may prove helpful. Akee et al. (2017) and Chetty et al. (2020) follow this approach, although they do not exploit the tax information in the Treasury data.

¹¹ We build on the procedures in Gale et al. (2022a, b) in several ways. We impose a \$10,000 cap on state and local tax deductions in tax year 2018, we separate the alimony and child support received using relevant demographic information from the survey, and we include non-filers. A key finding in Gale et al. (2022b) is that reported pass-through business income is twice as high in the SCF as it is in the IRS's Statistics of Income (SOI) data. For comparability to the IRS data, we cut the SCF-reported business income by half when estimating income tax liabilities.

Income (SOI) (the blue line) is small and relatively stable over time. Estimated revenues are \$1,517 billion in 2018, almost identical to the published IRS figure of \$1,510 billion. Appendix Table 1 shows that in data for tax year 2018, the simulated distributions of the number of returns and total income are relatively close to those in the SOI data. Income aggregates by income category are within 10 percent of published IRS figures in income groups that collectively account for 96 percent of total income.¹²

Since the 1998 survey, the SCF has asked respondents to describe themselves as white, Black or African American, Hispanic or Latino, Asian, American Indian or Alaska Native, Native Hawaiian or Pacific Islander, or other. Respondents can report more than one race but are asked which race they identify with most strongly, which we use as the race classifier.¹³ We assume that the spouse and respondent are the same race, thus allowing us to define couples as either Black or white.¹⁴

Our overall sample includes all married couples of all races and pools the eight SCF survey waves since 1998. Appendix Table 2 reports sample statistics. The raw count of tax units includes 19,014 with a white respondent and 1,182 with a Black respondent. The SCF generates

¹² The difference in number of tax units for those with AGI below \$50,000 arises because we do not include dependent filers. SOI (2019) reports 9.4 million dependent filers in 2018 with AGI under \$50,000. Our model generates 9.3 million fewer filers with an AGI under \$50,000 than the SOI data (including filers with an AGI of zero).

¹³ In sensitivity analysis, we find that restricting the sample to people solely reporting one race has virtually no effect on the results. A relatively small number of people report multiple racial identifications—about 6.8 percent of the sample in the 2019 survey, up from 2.3 percent in 2004. The public-use version of the SCF only provides information about whether a respondent reported identifying as more than one race, not what the other races are. Because questions about ethnicity were only asked starting in the 2004 survey, our sample of those who identify most strongly as Black or white may also include a small proportion of respondents of Hispanic or Latino culture or origin. In 2019, 90.2 percent of those who identified their ethnicity as Hispanic or Latino also reported it as their racial identification.

¹⁴ According to the 2010 Census, approximately 83.4 percent of married couples were comprised of two white spouses, 6.7 percent were comprised of two Black spouses, and 0.9 percent were comprised of one Black and one white spouse. Notably, Black men are more likely to have a racial intermarriage than Black women (U.S. Census Bureau, 2011).

five "implicates" for each unit to account for missing data. We employ information on all five implicates for each couple, generating a data set with about 95,000 white couples and almost 6,000 Black couples. We weight results using the SCF's replicate weights divided by eight (because we are using eight survey waves). Income and deductions are adjusted to 2018 dollars using the urban Consumer Price Index.

In each survey year, about half of white tax units and a quarter of Black units reported being married. About 78 percent of couples in the overall sample are white and 7 percent are Black, with the share among whites falling from 83 percent in 1998 to 75 percent in 2019, while the Black share was fairly constant (the remainder being other racial categories).

B. Measuring Marriage Bonuses and Penalties

We calculate the marriage penalty or bonus a couple faces as the difference between (a) their tax liability as a couple filing jointly and (b) their combined liability in the event they could both file as singles, or if children are present, as a single filer and a head of household.¹⁵ Many previous researchers have followed this approach.¹⁶ The key issue in these analyses is how to reorganize the family unit and its financial and living arrangements assuming the couple is no longer married.

Following Bull et al. (1999), we assume that married couples could duplicate the pooling of assets and expenses that occurs within their marriages without actually being married. For

¹⁵ In principle, measures of marriage bonuses and penalties can be derived by either estimating (1) the difference between what a sample of married couples would pay if they were not married or (2) the difference between what a sample of unmarried individuals would pay if they were married. As noted, we follow the first approach. Using the second approach requires either estimating the earnings of a single person's potential (but unobserved) spouse (Dickert-Conlin and Houser, 1998) or focusing on people who live with their partner (Alm and Leguizamon, 2015). It was beyond the scope of this paper to match unmarried individuals to one another and compute their tax liabilities as if they were married and filing a joint return. We explored the possibility of computing marriage penalties for cohabitating unmarried couples, but the SCF sample sizes were small.

¹⁶ Alm and Leguizamon, 2015; Alm and Whittington, 1996; Bull et al., 1999; Congressional Budget Office, 1997; Feenberg and Rosen, 1995.

income and expenses, we assume that the couple (and dependents) would continue to reside together and share expenses. Wages, self-employment income, pension income, Social Security benefits, unemployment benefits, and miscellaneous forms of earned income would be the same as when married and would be retained by the earner or beneficiary. Unearned income (dividends, interest, capital gains, etc.) and associated expenses would be the same as when married and would be divided evenly between the spouses. (See Appendix Table 3.)

Regarding tax provisions, we assume that the couple would retain the same exemptions (in pre-2018 law), eligible expenses for credits, and above-the-line and itemized deductions. We assume that the higher-income spouse would claim all above-the-line deductions and itemizable expenses. If there are no dependents, the couple would file as singles. If dependents are present, the higher-income spouse would file as head-of-household (if otherwise eligible) and claim all dependents. (See Appendix Table 4.)

These assumptions, consistent with earlier studies, generally hold taxpayer behavior constant with respect to marital status. For example, we do not incorporate any labor supply responses to marriage penalties and bonuses, consistent with Bull et al. (1999), Eissa and Hoynes (2000), Alm and Leguizamon (2015), and Alm et al. (2023). But we allow individuals to itemize deductions to reduce tax liability.

IV. RESULTS UNDER 2018 TAX LAW

<u>A. Results for Whole Sample</u>

Table 2 reports results for the whole sample, inclusive of all races. Overall, 43 percent of couples incurred a marriage penalty, averaging \$2,064 or 1.5 percent of AGI. Likewise, about 43 percent of couples received a bonus, averaging \$3,062 or 2.7 percent of AGI. For the remaining 14 percent of couples, marital status had a negligible effect (less than \$10) on annual income tax

liability.¹⁷ On average, couples paid \$432 less in income taxes, representing 0.4 percent of AGI, than if they had not been married.

B. Results by Race

Table 2 also presents summary results by race. In general, Black couples bear greater tax burdens from marriage than white couples do. Because Black couples' AGI is, on average, less than the income of white couples, the dollar amount of their marriage bonuses and penalties may also be lower, but the "penalty rate" or "bonus rate"—the penalty or bonus amount as a share of AGI—can be higher.

A greater proportion of Black couples than white couples faced marriage penalties: 46 percent compared to 43 percent. Among those facing penalties, the average penalty rate was higher for Black couples than white couples (1.8 percent versus 1.4 percent) but because of the differences in income between the groups, the average dollar amount for Black couples was lower than for white couples: \$1,804 compared to \$2,091.

Conversely, a smaller proportion of Black couples than white couples received marriage bonuses: 36 percent compared to 43 percent. Among those receiving bonuses, the average bonus rate was about the same: 2.6 percent for Black couples and 2.7 percent for white couples. This corresponds to average bonuses of \$1,926 for Black couples and \$3,304 for white couples.

Among all Black couples, the average tax effect of marriage was a penalty of \$148, corresponding to a net penalty rate of 0.2 percent of AGI. Among all white couples, the average tax effect of marriage was a bonus of \$514, corresponding to a net bonus rate of 0.4 percent.

C. Results by Race and Income

We now turn to results broken out by race and income. Figures 2 and 3 present results on

¹⁷ For approximately 30 percent of these couples, this was because they did not have any income tax liability if they filed jointly or as individuals.

the prevalence of marriage penalties and the average penalty as a share of income among those who face penalties. Figures 4 and 5 present analogous information for marriage bonuses. Figure 6 presents net average marriage bonuses (bonuses less penalties, as a share of income).

We break the sample into four income groups: couples with AGI under \$50,000, between \$50,000 and \$100,000, between \$100,000 and \$200,000, and above \$200,000.¹⁸ Within income groups, Brown (2021) posits that the incidence of marriage penalties would be greater for Black couples than for white couples. Our findings support this view. In AGI groups above \$50,000, the prevalence of marriage penalties was 8-13 percentage points higher for Black couples relative to white couples. The gap was only 1 percentage point for couples with AGI below \$50,000, in part because 31 percent of these couples faced no income tax liability whether they were married or not.

Among couples with penalties, the penalty as a share of income was consistently higher for Black couples than for white couples in all income groups, though the difference is negligible in the very top income group. Bonuses were generally more prevalent and a larger share of income for white couples—except in the very top income group.

The net bonus rates followed a U-shape pattern for both Black and white couples initially hovering around 1 percent for couples with income below \$50,000, declining between \$50,000 and \$200,000, and then rising to 0.4 percent for the highest-income Black couples and 0.7 percent for white couples. But Black couples with income between \$50,000 and \$200,000 faced a net penalty on average, whereas marriage had little net impact overall for white couples in those income groups.

D. Results by Number of Workers and Earnings Shares

¹⁸ Married couples with negative AGI constituted 0.4 percent of the sample and are excluded from the income groups but included in the aggregate figures.

Controlling for income, racial disparities in the tax treatment of marriage could occur for many reasons. One possible factor is differences in the relative earnings among spouses (including one- versus two-earner couples).

As described in Section II, controlling for income, couples with more equal spousal earnings are more likely to face penalties and to face larger penalties. As previously shown, Black couples were more likely to have two earners (Figure 1). And even among two-earner couples, relative spousal earnings were more equal for Black couples than white couples (Figure 7). For 59 percent of Black two-earner couples, but only 45 percent of white two-earner couples, the lower-earning spouse's labor compensation (wages, salaries, and self-employment income) was at least 60 percent of the compensation of the other spouse. The lower-earner's compensation was less than 40 percent of that of other spouse for a third of white two-earner couples but less than 20 percent of Black two-earner couples.

For these reasons, it follows that marriage penalties were more prevalent and a greater share of income (and bonuses were less prevalent and a smaller share of income) among twoearner Black couples than two-earner white couples. About 75 percent of Black two-earner couples and 69 percent of their white counterparts experienced marriage penalties—more than 25 percentage points higher for each two-earner group than the shares for each group overall.

Conversely, only 18 percent of two-earner Black couples and 26 percent of two-earner white couples received bonuses from marriage, with each of those shares at least 17 percentage points below the average for the respective racial group. Among two-earner couples, Black couples faced a net penalty rate of 1.2 percent of income (a percentage point higher than for the overall group), while white two-earner couples faced a net penalty of 0.5 percent of income

17

(compared to a bonus of 0.4 percent for the overall group).¹⁹

E. Results by Presence of Dependents

As described above, the presence of children can also contribute to marriage penalties and bonuses, due to the head-of-household filing status, the phase-ins of the child tax credit and the EITC, and the phaseout of the EITC. As previously shown, Black couples were significantly more likely to have children than white couples, controlling for income (Figure 1).

Marriage penalties were generally more prevalent and represented a greater share of income (and bonuses were less prevalent and represented a smaller share of income) among Black couples with dependents than among white couples with dependents. Among those with dependents, 62 percent of Black couples and 57 percent of white couples incurred marriage penalties—16 and 14 percentage points, respectively, higher than the overall averages by race. If the head-of-household filing status did not exist, the figures would fall to 46 percent and 41 percent, respectively.

The impact of having dependents varied significantly by income. Couples with dependents and income below \$50,000 had a markedly lower probability of facing a marriage penalty—31 percent for both Black and white couples—compared to couples with more than \$200,000 of income, where 49 percent of Black couples and 71 percent of white couples with dependents, faced penalties (possibly because the value of head-of-household filing status increases as people move into higher tax brackets). Low-income households also were more

¹⁹ We undertook a variety of forms of sensitivity analysis. First, changing the threshold below which a calculated marriage penalty or bonus? is deemed to be not a penalty or bonus? from \$10 to \$100 has virtually no effect on the results. Only 3-4 percent of those with bonuses or penalties experience a change below \$100 in absolute value. Second, restricting the sample to couples with respondents between the ages of 25 and 64 slightly raises the overall prevalence of penalties – to 46 percent from 43 percent—but only because it increases the percentage of the sample represented by two-earner couples. The prevalence or marriage penalties for one-earner couples and for two-earner couples changed only by 1 percentage point. Third, allocating capital income in the same proportion as labor income (as opposed to splitting it equally among spouse) has very small effects – 42 percent of couples face penalties, 44 percent receive bonuses, compared to 43 percent and 43 percent, respectively, in the base case.

likely to receive bonuses than most other households.

Black couples with dependents faced an average net marriage penalty of 0.7 percent of income; white couples with dependents obtained, on average, a net marriage bonus of 0.2 percent of income, increasing the gap between Black and white couples to 0.9 percentage points compared to 0.5 percentage points for all couples regardless of the presence of dependents.²⁰

F. Regression Analysis

To analyze these findings more formally and to understand the relative contribution of each characteristic for the presence and size of marriage penalties, we turn to regressions. The sample contains all Black and white couples. The first specification controls for the SCF year, AGI class, the lower earner's labor compensation as a fraction of the higher earner's compensation (earnings ratio, ER), number of dependents (Dep), and the race of the respondent, and is given by:

$$X = \alpha_1 + \sum_{k=2}^{8} \alpha_{2k} Y ear_k + \sum_{j=2}^{4} \beta_j A G I_{ij} + \gamma_1 E R_i + \gamma_2 D e p_i + \gamma_3 Race_i + \varepsilon$$
(1)

where i indexes couples, j indexes AGI class (with < \$50,000 omitted), and k indexes SCF waves.

The second specification incorporates interactions between income and the other variables:

²⁰ Repealing the EITC would reduce marriage penalties and increase marriage bonuses significantly – to an average net bonus of 0.9 percent of income from a net bonus of 0.4 percent of income in the base case above. Married couples with two earners would, on average, receive a net bonus rate of 0.1 percent rather than a net penalty of 0.6 percent; couples with dependents would receive a higher net bonus rate – on average, 1.1 percent up from 0.2 percent. Eliminating the EITC would shift Black couples from net penalties on average (0.2 percent) to net bonuses on average (0.7 percent). Interestingly, eliminating the EITC also causes a small reduction in marriage penalties for high-income couples. Among high-income couples, a spouse with very low earnings cannot receive the EITC if filing jointly but may qualify for EITC for workers who do not live with children, if filing as an individual. This tends to increase the prevalence of marriage penalties, even among higher-income couples, but has a very small impact on the magnitude of the penalty since the amount of the "childless EITC" is much smaller than the credit received by those with children.

$$X = \alpha_{1} + \sum_{k=2}^{8} \alpha_{2k} Y ear_{k} + \sum_{j=2}^{4} \beta_{j} AGI_{ij} + \sum_{j=1}^{4} \gamma_{1i} AGI_{ij} ER_{i} + \sum_{j=1}^{4} \gamma_{2i} AGI_{ij} Dep_{i} + \sum_{j=1}^{4} \gamma_{1i} AGI_{ij} Race_{i} + \varepsilon$$
(2)

Table 3 reports results for the prevalence of marriage penalties. In the linear probability model, X = 1 if the couple faces a marriage penalty, 0 otherwise. The results in the first column indicate that marriage penalties are more prevalent in income classes above \$50,000, when spousal earnings are relatively equal and when dependents are present. Controlling for these factors, the coefficient on the indicators for race is estimated to be zero.

The second column reports results from estimating (2). The same general coefficient patterns occur as in the first specification, for the variables other than race, except that the presence of dependents in the highest income class appears to reduce the prevalence of marriage penalties. The regressions suggest that, even after controlling for other variables, Black couples are more likely to face marriage penalties than other couples. This suggests that other factors besides those presented above affect the creation of marriage penalties or bonuses.

Table 4 reports results from OLS regressions that set X equal to the value of the couple's marriage bonus (+) or penalty (-). In this first specification, bonuses rise with income class, fall with more equal spousal earnings and with dependents, and race has no impact. After interacting the explanatory variables with income, the same patterns emerge for the variables other than race. But Black couples face larger penalties in income classes above \$50,000, even after controlling for the other characteristics.

V. ALTERNATIVE APPROACHES

The tax treatment of marriage has changed over time, reflecting shifting views about what constitutes equitable treatment. The income tax that took effect in 1913 was imposed on individuals rather than families (Bull et al, 1999; Michelmore, 2018). But couples in communityproperty states soon discovered that they could reduce their combined tax liability by strategically shifting income between spouses. This led to protests by residents in other states. In response, Congress created the joint filing status in 1948 and widened the tax brackets for married couples to be twice the width for single filers. This eliminated marriage penalties but increased the prevalence of marriage bonuses (singles penalties), particularly for one-earner couples.

Unmarried individuals then protested, leading Congress to make the tax code more favorable to that group, by creating the head-of-household filing status in 1951 and by changing the tax brackets in 1969. These changes, however, increased the likelihood of marriage penalties, particularly for two-earner couples. To address that concern, Congress enacted a two-earner deduction in 1981 but repealed it two years later. In response to continuing concerns about equity, Congress took several incremental steps between 2001 and 2017 to reduce marriage penalties for two-earner couples, but the broad scope of those tax changes also had the effect of boosting bonuses for many one-earner couples. (See Appendix Table 5 and the text below for details.)

These prior discussions notably omitted consideration of racial dimensions of the tax treatment of marriage. In this section, we examine how legislative changes since 2000 and two policy reforms affect Black and white couples differently.

A. Impact of Legislation Since 2000

The Economic Growth and Tax Reconciliation Relief Act (EGTRRA) of 2001 reduced

21

marriage penalties and increased marriage bonuses by raising the standard deduction and the income thresholds for the two lowest tax brackets for couples to twice the levels for single filers (Kiefer et al, 2002). Both EGTRRA and The American Recovery and Reinvestment Act of 2009 stretched the EITC phaseout range for married couples, although stopping short of setting the credit thresholds for married couples at double the levels for unmarried filers. In TCJA in 2017, Congress extended (through 2025) the broader rate brackets for married couples to all but those in the top two rate brackets (El-Sibaie 2018).

Other provisions in post-2000 tax legislation indirectly impacted marriage penalties and bonuses. For example, EGTRRA and TCJA reduced marginal tax rates, which generally lowers marriage penalties, though it could snag high-income taxpayers on the alternative minimum tax. EGTRRA doubled the maximum child tax credit but increased marriage penalties because the beginning of the phaseout of the enhanced credit for couples remained at \$110,000—less than twice the threshold of \$75,000 for unmarried filers. TCJA doubled the maximum credit again and removed the associated marriage penalty by increasing the phaseout thresholds to \$200,000 for unmarried filers and \$400,000 for married couples through 2025.²¹ A variety of other provisions enacted since 2000 had smaller effects on marriage penalties.

Figures 8-10 show how the evolution of tax law affected marriage penalty patterns.²²

²¹ For 2021 only, the American Rescue Act expanded the child tax credit in ways that affected both marriage bonuses and penalties. First, the act made the credit fully refundable, allowing claimants to get the full credit even if their income was zero. Doing so eliminated marriage bonuses because the amount of the enhanced credit did not increase as income initially rose. The act also raised the amount of the credit but then phased out the incremental increase when income rose above \$150,000 for married couples, \$112,500 for heads of households, and \$75,000 for single filers. Because the threshold for heads of households was not half the level for joint filers, a couple incurred a marriage penalty in that range. The remaining credit would still phase out at the TCJA levels, which did not result in penalties because the threshold for married couples was twice the level for all unmarried filers, whether they filed as a single or head of household. President Biden has proposed restoring the 2021 expansion but just through the end of 2025 except for full refundability which would become permanent under his plan.

²² Prior to 2017, the itemized deduction for state and local taxes (SALT) was unlimited. In 2017, TCJA limited the deduction to \$10,000. As a result, there are no SOI data for 2018 on what taxpayers would have deducted had the

Several aggregate trends are worth highlighting. First, the share of couples who faced marriage penalties fell from 56 percent under 2000 law to 47 percent under 2015 law and then to 44 percent under 2018 law. Second, differences in the prevalence of marriage penalties between Black couples and white couples increased slightly over this period (Figure 8). Third, the share of couples receiving bonuses rose significantly due to tax law changes, from 28 percent and 32 percent of Black couples and white couples, respectively, under 2000 law, to 35 percent and 43 percent under 2018 law (Figure 9). Fourth, couples incurred a net penalty rate of 0.9 percent under 2000 law, which fell to 0.1 percent under 2015 law. The change to 2018 law did not alter the net penalty rate for Black couples but gave white couples a net bonus rate of 0.4 percent (Figure 10). That is, TCJA eliminated marriage penalties, on average, for white couples but had no discernable effect, on average, for Black couples.

There were somewhat divergent trends by income group. Among couples with AGI below \$50,000, the change in law from 2000 to 2015 greatly reduced the prevalence of marriage penalties and increased the prevalence of marriage bonuses. Among high-income households, both the changes between 2000 and 2015 law and the changes between 2015 and 2018 law significantly reduced marriage penalties and boosted bonuses. TCJA had a particularly strong effect for couples with AGI above \$200,000, raising the net marriage bonus rate by 1.0 percent of income for Black couples and 1.3 percent of income for white couples (Figure 10).

B. Optional Individual Filing

As noted above, the two major drivers of marriage penalties, besides income, are the

deduction been unlimited. This makes it difficult to calculate SALT under 2000 and 2015 law (when the state and local tax deduction was unlimited) using the 2019 SCF. Therefore, to do the comparisons with prior law, we dropped the 2019 SCF data from the sample. Thus, unlike our earlier analysis, the estimates for prior law and 2018 law in this section are based on seven waves of the SCF through 2016. The estimates of the prevalence and magnitude of marriage bonuses and penalties under 2018 law, using the truncated data, are similar to those using the full sample and reported earlier. Our analysis of the 2015 and 2018 laws do not reflect the premium tax credit, enacted as part of the Affordable Care Act.

presence of dependents and the existence of two-earner families. We examine reforms that could at least partially address these issues. The first reform, analyzed in this section, would give spouses the choice to file as a married couple, as singles, or (if children are present) head of household. This would help address the role of dependents by allowing couples to use the headof-household status and to avoid marriage penalties in the EITC and child tax credit.

To be clear, such a change would raise difficult issues. It would be a substantive change to the tax code with implications for horizontal and vertical equity (Liebman and Ramsey, 2019; Rosen, 1977). It would require rules to allocate capital income, deductions, and dependents, and some couples might choose to forgo the tax cut rather than deal with the additional complexity involved in filing tax returns. And even with statutory allocation rules, creative couples probably would still find ways to shift certain forms of capital income to the partner facing the lower marginal tax rate, creating marriage subsidies—or, more pointedly, singles penalties (Eickmeyer et al., 2019). These issues notwithstanding and assuming that the statutory allocation rules would correspond to those we made above (in Appendix Table 3), we estimate the effects of moving to an optional individual-filing system²³.

The policy would eliminate *all* marriage penalties, thus reducing taxes for 46 percent of Black couples and 43 percent of white couples. It would have *no* effect on previously existing marriage bonuses. The annual cost would have been about \$49 billion in 2018. Figure 11 shows how this policy would change net average bonuses as a share of income, relative to 2018 law.

²³ An alternative option would be to mandate individual filing, We estimate that requiring all couples to file individually would generate a net tax increase of \$21.6 billion and eliminate aggregate marriage penalties of \$42.2 billion and aggregate marriage bonuses of \$63.8 billion, In addition to increasing taxes on couples with bonuses, mandating individual filing would not enable other taxpayers to opt out of individual filing if they found the additional complexity of allocating income and expenses between the spouses imposed a greater burden (time costs, payments to tax advisers) than the actual penalty. (In our estimates of optional individual filing, we assume that all couples with penalties would choose to file as unmarried taxpayers if given the choice,)

Black and white couples would experience an increase in the net bonus rate of 1.1 percent of income and 0.7 percent of income, respectively. Among couples with dependents, the effect would be larger—Black couples would receive a change of 1.8 percent of income; white couples would receive a change of 0.9 percent of income. For both races, the increase in bonuses declines as a share of income as income rises. That is, the policy change would be progressive among married couples.

Although not shown in Figure 11, the policy change would give tax cuts to a greater share of white tax units and adults (22 percent and 30 percent, respectively) than Black tax units and adults (10 percent and 17 percent, respectively).

C. Two-Earner Deduction

Allowing couples to deduct a portion of the lower-earning spouse's income would help reduce the impact on marriage penalties of having two earners (Piggott and Whalley, 1996). From 1981 to 1983, two-earner couples could subtract 10 percent of the lower-earner's earnings up to \$30,000, for a maximum deduction of \$3,000, from AGI. We examine a simplified variant of that policy, which would have allowed taxpayers to deduct 10 percent of the lower earner's earnings up to \$90,000 in 2018—roughly the same threshold as under 1981 law, adjusted for inflation. Under our variant, however, the deduction would apply to taxable income rather than AGI; as a consequence, the option would not impact eligibility for the lower tax rates on capital gains, the EITC, and various provisions that are explicitly linked directly to AGI.²⁴

The two-earner deduction would have reduced marriage penalties by \$14.8 billion in

²⁴As an illustrative calculation, suppose the lower-earning spouse earned 50 percent as much as the higher earner (and earned less than \$90,000), and the family had only labor income. Then the lower-earning spouse would have earned 33.3 percent of family income. A 10 percent deduction would reduce family income by 3.33 percent. If the family faced a 15 percent (25 percent) marginal federal income tax rate, tax liabilities would fall by about 0.50 percent (0.83 percent) of income. This figure can be compared to the average marriage penalty rates reported in the text and the effects of the policy reported below.

2018 and raised marriage bonuses by \$4.8 billion, reducing overall revenues by \$19.6 billion. The policy would reduce the prevalence of marriage penalties by only 2.0 percentage points for Black couples and 3.3 percentage points for white couples (Figure 12). The effects are somewhat larger among two-earner couples—4.9 percentage points and 7.0 percentage points, respectively—but still small. Figure 13 shows somewhat larger effects on the prevalence of bonuses.

What stands out in Figures 12 and 13 is how much the reform would benefit higherincome households. Among couples with income above \$200,000, the prevalence of penalties would fall by 11.3 and 8.5 percentage points for Black couples and white couples, respectively and 17.4 percentage points and 15.1 percentage points for two-earner couples in those groups. Figure 14 shows that the impact of the reform on net marriage bonuses as a share of income is larger for Black couples than whites and is particularly large among high-income couples. The value of the deduction rises with income because higher-income couples are more likely to be able to claim the maximum deduction of \$9,000 and because the value of deductions increases as the taxpayer moves into higher tax brackets. As a result, the reform would be regressive among married couples.

The two-earner deduction has a greater impact on the number of affected tax units and adults than on the prevalence of marriage penalties for two reasons. First, a two-earner couple who faced a penalty under current law may still be subject to a penalty even with the deduction. Second, all two-earner couples with positive tax liability—including those who had marriage bonuses under current law—could receive the deduction. For that reason, we also observe a greater share of tax units and adults receiving a tax cut under the two-earner deduction than under the option to file individually.

26

Although not shown in Figure 12, the reform would give tax cuts to a greater share of white tax units and adults (25 percent and 33 percent, respectively) than Black tax units and adults (13 percent and 20 percent, respectively).

VI. CONCLUSION

We provide new evidence on marriage penalties by race using the 1998-2019 Survey of Consumer Finances, coupled with methods that convert SCF household units into tax units, methods for determining the tax liability of married couples if they were unmarried, and the application of NBER's TAXSIM. We find that Black couples are more likely than white couples to experience an income tax penalty from marriage and to face higher penalties. We show that these patterns arise because, controlling for income, Black spouses have more equal earnings than white spouses, as Brown (2021) hypothesizes, and because Black couples are more likely to have dependents.

There is no perfect solution to the tax treatment of marriage, and the byzantine system that has emerged reflects policymakers' struggles to reconcile a variety of conflicting goals. Our findings, along with Brown (2021), Alm et al. (2023) and other emerging research, suggest the need to add one more consideration into the discussion: the impact on racial equity.

Several caveats apply to our analysis. First, we focus solely on annual measures of marriage penalties and bonuses, but the effects could persist over time (Ilin et al., 2022). Second, we focus on the federal individual income tax, but marriage penalties and bonuses can occur in other taxes and in federal and state benefit programs as well (Steuerle, 2023). Third, data limitations force us to omit certain key features of the income tax, such as the premium tax credit.

Finally, we do not investigate the underlying economic and social causes of the racial

disparities in the tax treatment of marriage. The differential marriage tax that is imposed on Black couples relative to white couples can be related to issues of structural racism—as both the result and the cause of a system of reinforcing disparities. For example, low earnings and high incarceration rates among Black men—for reasons related to structural racism—may have driven more Black women into the labor force. In turn, the greater prevalence of marriage penalties among Black couples reduces their after-tax income relative to white couples with the same pretax income. Future research could fruitfully focus on empirical investigations of other aspects of the tax code, and public policy more generally, that are race-blind on the surface but are both the cause and effect of racial disparities and structural racism.

REFERENCES

- Akee, Randall, Maggie Jones, and Sonya Porter. 2017. "Race Matters: Income Shares, Income Inequality, and Income Mobility for All U.S. Races." w23733. Cambridge, MA: National Bureau of Economic Research. https://doi.org/10.3386/w23733.
- Alm, James, and J. Sebastian Leguizamon. 2015. "Whither the Marriage Tax?" *National Tax Journal* 68 (2): 251–79. https://doi.org/10.17310/ntj.2015.2.02.
- Alm, James, Sebastian Leguizamon, and Susane Leguizamon.2023. "Race, Ethnicity, and Taxation of the Family: The Many Shades of the Marriage Tax/Subsidy." *National Tax Journal*.
- Alm, James, and Leslie A. Whittington. 1995. "Does the Income Tax Affect Marital Decisions." *National Tax Journal* 48 (4): 565–72.
- Alm, James, and Leslie A. Whittington. 1996. "The Rise and Fall and Rise... Of the Marriage Tax." *National Tax Journal* 49 (4): 571–89. https://doi.org/10.1086/NTJ41789227.
- Alm, James, and Leslie A. Whittington. 1999. "For Love or Money? The Impact of Income Taxes on Marriage." *Economica* 66 (263): 297–316. https://doi.org/10.1111/1468-0335.00172.
- Alm, James, and Leslie A. Whittington. 2003. "Shacking Up or Shelling Out: Income Taxes, Marriage, and Cohabitation." *Review of Economics of the Household* 1 (3): 169–86. https://doi.org/10.1023/A:1025093300161.
- Bearer-Friend, Jeremy. 2019. "Should the IRS Know Your Race? The Challenge of Colorblind Tax Data." *Tax Law Review* 73 (1): 1–68.
- Bittker, Boris I. 1975. "Federal Income Taxation and the Family." *Stanford Law Review* 27 (6): 1389. https://doi.org/10.2307/1228181.

- Brown, Dorothy A., 1997a. "The Marriage Bonus/Penalty in Black and White." *University of Cincinnati Law Review* 65 (3), 787-798.
- Brown, Dorothy A., 1997b. "Race, Class, and Gender Essentialism in Tax Literature: The Joint Return." *Washington & Lee Law Review* 54 (4), 1469-1512.
- Brown, Dorothy A., 1999a. "Racial Equality in the Twenty-first Century: What's Tax Policy Got To Do With It?" *University of Arkansas at Little Rock Law Review* 21 (4), 759-768.
- Brown, Dorothy A., 1999b. "The Marriage Penalty/Bonus Debate: Legislative Issues in Black and White." *New York Law School Journal of Human Rights* 16 (1), 287-302.
- Brown, Dorothy A., 2005. "Tax Treatment of Children: Separate But Unequal." *Emory Law Journal* 54 (2), 755-842.
- Brown, Dorothy A. 2021. The Whiteness of Wealth: How the Tax System Impoverishes Black Americans – and How We Can Fix It. New York: Crown.
- Bull, Nicholas, Janet Holtzblatt, James R. Nunns, and Robert Rebelein. 1999. "Defining and Measuring Marriage Penalties and Bonuses." *The Department of Treasury Office of Tax Analysis*. OTA Paper 82-Revised.
- Chetty, Raj, Nathaniel Hendren, Maggie R Jones, and Sonya R Porter. 2020. "Race and Economic Opportunity in the United States: An Intergenerational Perspective." *The Quarterly Journal of Economics* 135 (2): 711–83. https://doi.org/10.1093/qje/qjz042.
- Congressional Budget Office. 1997. "For Better or for Worse: Marriage and the Federal Income Tax." https://www.cbo.gov/sites/default/files/105th-congress-1997-1998/reports/marriage.pdf.

Cronin, Julie-Anne, Portia DeFilippes, and Robin Fisher. 2023. "Tax Expenditures by Race and

Hispanic Ethnicity: An Application of the U.S. Treasury Department's Race and Hispanic Ethnicity Imputation." *The Department of Treasury Office of Tax Analysis*. Working Paper 122. https://home.treasury.gov/system/files/131/WP-122.pdf

- Dickert-Conlin, Stacy, and Scott Houser. 1998. "Taxes and Transfers: A New Look at the Marriage Penalty." *National Tax Journal* 51 (2): 175–217. https://doi.org/10.1086/NTJ41789323.
- Eickmeyer, Kasey J., Wendy D. Manning, and Susan L. Brown. 2019. "What's Mine Is Ours? Income Pooling in American Families." *Journal of Marriage and Family* 81 (4): 968–78. https://doi.org/10.1111/jomf.12565.
- Eissa, Nada, and Hilary Williamson Hoynes. 2000. "Explaining the Fall and Rise in the Tax Cost of Marriage: The Effect of Tax Laws and Demographic Trends, 1984-97." *National Tax Journal* 53 (3.2): 683–711. https://doi.org/10.17310/ntj.2000.3S.04.
- El-Sibaie, Amir. 2018. "Marriage Penalties and Bonuses under the Tax Cuts and Jobs Act." No. 573.

Fiscal Fact. Tax Foundation. https://taxfoundation.org/tax-cuts-and-jobs-act-marriage-penalty/.

Fisher, Robin. 2023. "Estimation of Race and Ethnicity by Re-Weighting Tax Data." The Department of Treasury Office of Tax Analysis. Technical Working Paper 11. https://home.treasury.gov/system/files/131/TP-11.pdf

Feenberg, Daniel, and Elisabeth Coutts. 1993. "An Introduction to the TAXSIM Model." Journal of Policy Analysis and Management 12 (1): 189–94. https://doi.org/10.2307/3325474.

Feenberg, Daniel R., and Harvey S. Rosen. 1995. "Recent Developments in the Marriage Tax."

National Tax Journal 48 (1): 91-101. https://doi.org/10.1086/NTJ41789126.

- Gale, William G. 2021. "Public Finance and Racism." *National Tax Journal* 74 (4): 953–74. https://doi.org/10.1086/717146.
- Gale, William, Swati Joshi, Christopher Pulliam, and John Sabelhaus. 2022a. "Simulating Income Tax Liabilities in the Survey of Consumer Finances." *Urban-Brookings Tax Policy Center*. https://www.brookings.edu/research/simulating-income-tax-liabilities-in-the-surveyof-consumer-finances/.
- Gale, William, Swati Joshi, Christopher Pulliam, and John Sabelhaus. 2022b. "Taxing Business Incomes: Evidence from the Survey of Consumer Finances." *Urban-Brookings Tax Policy Center*. https://www.brookings.edu/research/simulating-income-tax-liabilities-in-the-surveyof-consumer-finances/.
- Howard, Kimberly, and Richard V. Reeves. 2014. "The Marriage Effect: Money or Parenting?" Washington, D.C.: The Brookings Institution. https://www.brookings.edu/research/themarriage-effect-money-or-parenting/.
- Ilin, Elias, Laurence Kotlikoff, and Melinda Pitts. 2022. "Is Our Fiscal System Discouraging Marriage? A New Look at the Marriage Tax." w30159. Cambridge, MA: National Bureau of Economic Research. https://doi.org/10.3386/w30159.
- Kearney, Melissa, and Phillip Levine. 2017. "The Economics of Non-Marital Childbearing and the 'Marriage Premium for Children." w23230. Cambridge, MA: National Bureau of Economic Research. https://doi.org/10.3386/w23230.
- Khitatrakun, Surachi, Gordon B. Mermin, Benjamin R. Page, and Jeffrey Rohaly. 2023. "A New Approach for Estimating the Impact of Tax Policies by Race and Ethnicity." Urban-Brookings Tax Policy Center. https://www.taxpolicycenter.org/publications/new-approach-

estimating-impact-tax-policies-race-and-ethnicity/full'

- Kiefer, Donald, Robert Carroll, Janet Holtzblatt, Allan Lerman, Janet McCubbin, David Richardson, and Jerry Tempalski. 2002. "The Economic Growth and Tax Relief Reconciliation Act of 2001: Overview and Assessment of Effects on Taxpayers." *National Tax Journal* 55(1): 89–117. https://doi.org/10.17310/ntj.2002.1.06
- Liebman, Jeffrey, and Daniel Ramsey. 2019. "Independent Taxation, Horizontal Equity, and Return-Free Filing." *Tax Policy and the Economy* 33 (January): 109–30.
- McLanahan, Sara, and Isabel Sawhill. 2015. "Marriage and Child Wellbeing Revisited: Introducing the Issue." *The Future of Children* 25 (2): 3–9.
- Michelmore, Molly. 2018. "Creating the Marriage Penalty: Tax Politics, Gender, and Political Realignment in 1970s America." *Journal of Women's History* 30 (2): 136–60. https://doi.org/10.1353/jowh.2018.0017.
- Moran Beverly I. and William Whitford. 1996. "A Black Critique of the Internal Revenue Code." *Wisconsin Law Review*. 751-820.
- Piggott, John, and John Whalley. 1996. "The Tax Unit and Household Production." Journal of Political Economy 104 (2): 398–418.
- Ribar, David C. 2015. "Why Marriage Matters for Child Wellbeing." *The Future of Children* 25 (2).
- Rosen, Harvey S. 1987. "The Marriage Tax Is Down But Not Out." *National Tax Journal* 40 (4): 567–75.
- Shapiro, Thomas, Tatjana Meschede, and Sam Osoro. 2013. "The Roots of the Widening Racial Wealth Gap: Explaining the Black-White Economic Divide." Research and Policy Brief.Brandeis University Institute on Assets and Social Policy.

- Sjoquist, David L., and Mary Beth Walker. 1995. "The Marriage Tax and the Rate and Timing of Marriage." *National Tax Journal* 48 (4): 547–58. https://doi.org/10.1086/NTJ41789170.
- Steuerle, Eugene. 2023. "Does Government Policy Toward Marriage Make Sense?" June 18. https://governmentwedeserve.substack.com/p/does-government-policy-towardmarriage?utm_source=profile&utm_medium=reader2.
- U.S. Census Bureau. 2011. "Families and Living Arrangements, Table MS-3. Interracial Married Couples: 1980 to 2002," and unpublished data. https://www2.census.gov/library/publications/2011/compendia/statab/131ed/tables/12s0060. xls
- Wilcox, W. Bradford, Robert I. Lerman, and Joseph Price. 2015. "Mobility and Money in U.S.States: The Marriage Effect." Washington, D.C.: The Brookings Institution.
| | Married Couple | Two Singles | Two Singles | Two Singles |
|-----------------------------|-----------------|--------------------------|------------------|-------------|
| Total Income | 60,000 | 60,000 | 60,000 | 60,000 |
| Partner A's Income | N/A | 30,000 | 10,000 | 0 |
| Partner B's Income | | 30,000 | 50,000 | 60,000 |
| | | Simple Tax System | | |
| Combined Tax Owed | 6,000 | 2,000 | 4,000 | 6,000 |
| Penalty (-) or Bonus
(+) | | -4,000 | -2,000 | 0 |
| | Simple Tax Syst | tem with Brackets Cut in | Half for Singles | |
| Combined Tax Owed | 6,000 | 6,000 | 7,000 | 9,000 |
| Penalty (-) or Bonus
(+) | | 0 | +1,000 | +3,000 |

Table 1: Examples of Marriage Penalties and Bonuses Under Simple Tax System²⁵

²⁵ In the Simple Tax System, the first \$20,000 of income is tax-free, the next \$20,000 is subject to a marginal tax rate of 10%, and all additional income is subject to a marginal tax rate of 20%. There are no deductions, credits, exemptions, or dependents. Singles and married couples face the same tax schedule. When the tax brackets are cut in half for singles, they receive the first \$10,000 of income tax-free and pay 10% on the next \$10,000 of income and 20% on income above \$20,000.

Table 2: Summary Results for Married Couples

	All Couples	Black Couples	White Couples
Percent with penalty	43	46	43
Percent with bonus	43	36	43
Percent with neither penalty nor bonus	14	18	14
Average penalty among those with penalties (\$)	2,064	1,804	2,091
Average bonus among those with bonuses (\$)	3,062	1,962	3,304
Net average penalty (-) or bonus (+) (\$)	432	-148	514
Average penalty rate among those with penalties (Percent of AGI)	1.5	1.8	1.4
Average bonus rate among those with bonuses (Percent of AGI)	2.7	2.6	2.7
Net average penalty (-) or bonus (+) rate (Percent of AGI)	0.4	-0.2	0.4

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the SCF (1998 - 2019), and population weights are accordingly divided by eight.

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in the counts.

Variables	Specification 1	Specification 2
AGI: \$50,000 – 100,000	0.17***	0.14***
AGI: \$100,000 – 200,000	0.17***	0.11***
AGI: More than 200,000	0.15***	0.20***
Earnings Ratio	0.65***	
Has Dependents	0.15***	
Black Couple	0.00	
Earnings Ratio * AGI \$0 – 50,000		0.65***
Earnings Ratio * AGI \$50,000 – 100,000		0.52***
Earnings Ratio * AGI \$100,000 – 200,000		0.72***
Earnings Ratio * AGI More than \$200,000		0.82***
Has Dependents * AGI \$0 – 50,000		0.11***
Has Dependents * AGI \$50,000 – 100,000		0.23***
Has Dependents * AGI \$100,000 – 200,000		0.18***
Has Dependents * AGI More than \$200,000		-0.05***
Black * AGI \$0 – 50,000		-0.04***
Black * AGI \$50,000 – 100,000		0.02
Black * AGI \$100,000 – 200,000		0.03***
Black * AGI More than \$200,000		0.06***
Constant	0.06***	0.08***
Observations	100,107	100,107
R-Squared	0.32	0.33

Table 3: Determinants of the Prevalence of Marriage Penalties (Linear Probability Model)

*** p<0.01, **p<0.05, *p<0.1

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the Survey of Consumer Finances (1998 – 2019), and population weights are accordingly divided by eight.

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in the counts. We calculate the earnings ratio as the earnings of the lower-earning spouse as a percent of the higher-earning spouse's earnings, with values between 0 and 1. Earnings are defined as the sum of wages, salaries, and net business income, and couples with negative income (including business losses) are included in the "No-Earner" group. Earnings ratios for no-earner and one-earner couples were set at zero. Couples with negative AGI were dropped from the sample. Indicators for survey waves were included in the regression but are excluded for readability.

Variables	Specification 1	Specification 2
AGI: \$50,000 – 100,000	909***	862***
AGI: \$100,000 – 200,000	1,267***	2,194***
AGI: More than 200,000	5,635***	8,323***
Earnings Ratio	-4,958***	
Has Dependents	-201***	
Black Couple	-58*	
Earnings Ratio * AGI \$0 – 50,000		-1,465***
Earnings Ratio * AGI \$50,000 - 100,000		-2,274***
Earnings Ratio * AGI \$100,000 – 200,000		-5,006***
Earnings Ratio * AGI More than \$200,000		-17,364***
Has Dependents * AGI \$0 – 50,000		21
Has Dependents * AGI \$50,000 – 100,000		-826***
Has Dependents * AGI \$100,000 – 200,000		-975***
Has Dependents * AGI More than \$200,000		1,631***
Black * AGI \$0 – 50,000		27.59
Black * AGI \$50,000 – 100,000		-138***
Black * AGI \$100,000 – 200,000		-213***
Black * AGI More than \$200,000		-807**
Constant	807***	331***
Observations	100,107	100,107
R-Squared	0.14	0.21

*** p<0.01, **p<0.05, *p<0.1

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the Survey of Consumer Finances (1998 – 2019), and population weights are accordingly divided by eight.

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in the counts. We calculate the earnings ratio as the earnings of the lower-earning spouse as a percent of the higher-earning spouse's earnings, with values between 0 and 1. Earnings are defined as the sum of wages, salaries, and net business income, and couples with negative income (including business losses) are included in the "No-Earner" group. Earnings ratios for no-earner and one-earner couples were set at zero. Couples with negative AGI were dropped from the sample. Indicators for survey waves were included in the regression but are excluded for readability.

Figure 1: Characteristics of Couples by Race



Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the SCF (1998 - 2019), and population weights are accordingly divided by eight.

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.



Figure 2: Prevalence of Marriage Penalties, 2018 Law

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the SCF (1998 - 2019), and population weights are accordingly divided by eight.

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.



Figure 3: Average Penalty Rates Among Couples with Penalties, 2018 Law

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the SCF (1998 - 2019), and population weights are accordingly divided by eight.

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.



Figure 4: Prevalence of Marriage Bonus, 2018 Law

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the SCF (1998 - 2019), and population weights are accordingly divided by eight.

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.



Figure 5: Average Bonus Rates Among Couples with Bonuses, 2018 Law

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the SCF (1998 - 2019), and population weights are accordingly divided by eight.

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.



Figure 6: Net Marriage Bonus (+) or Penalty (-) as a Share of Income, 2018 law

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the SCF (1998 - 2019), and population weights are accordingly divided by eight.

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.



Figure 7: Lower Earner's Income as Share of Higher Earner's Income, Two-Earner Couples

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the SCF (1998 - 2019), and population weights are accordingly divided by eight.

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.



Figure 8: Prevalence of Marriage Penalties, 2000, 2015, 2018 Law

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. The data are from a pooled sample of seven waves of the SCF (1998 – 2016), and population weights are accordingly divided by seven.

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.



Figure 9: Prevalence of Marriage Bonuses, 2000, 2015, 2018 Law

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. The data are from a pooled sample of seven waves of the SCF (1998 – 2016), and population weights are accordingly divided by seven.

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.



Figure 10: Net Marriage Bonus (+) or Penalty (-) as a Share of Income, 2000, 2015, 2018 Law

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. The data are from a pooled sample of seven waves of the SCF (1998 – 2016), and population weights are accordingly divided by seven.

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.



Figure 11: Change in Net Marriage Bonus (+) or Penalty (-) as a Share of Income, Individual Filing Option

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the SCF (1998 - 2016), and population weights are accordingly divided by eight.

Notes: Under the individual filing option, couples could choose to file jointly or as single (or head of household, if qualified). Change is relative to 2018 law. Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.



Figure 12: Change in Prevalence of Marriage Penalties, Two-Earner Deduction

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the SCF (1998 - 2016), and population weights are accordingly divided by eight.

Notes: Under the policy, married couples could choose to deduct from taxable income 10 percent of the lower earner's earnings up to \$90,000, for a maximum deduction of \$9,000. Change is relative to 2018 law. Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.



Figure 13: Change in Prevalence of Marriage Bonus, Two-Earner Deduction

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the SCF (1998 - 2016), and population weights are accordingly divided by eight.

Notes: Under the policy, married couples could choose to deduct from taxable income 10 percent of the lower earner's earnings up to \$90,000, for a maximum deduction of \$9,000. Change is relative to 2018 law. Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.



Figure 14: Change in Net Marriage Bonus (+) or Penalty (-) as a Share of Income, Two-Earner Deduction

Source: Survey of Consumer Finances (SCF). Authors' calculations are derived using TAXSIM 35. Data are from a pooled sample of eight waves of the SCF (1998 - 2016), and population weights are accordingly divided by eight.

Notes: Under the policy, married couples could choose to deduct from taxable income 10 percent of the lower earner's earnings up to \$90,000, for a maximum deduction of \$9,000. Change is relative to 2018 law. Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as an individual whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. Dependents were restricted to those age 18 and under. Couples with negative adjusted gross income are excluded from the lowest-income group but are included in totals.

	Number	Number of Returns (Millions)			Aggregate Income Subject to Tax (Millions)		
Adjusted Gross Income Group	SCF	SOI	Ratio of SCF/SOI	SCF	SOI	Ratio of SCF/SOI	
(Thousands)							
Less than 25	40.9	50.4	0.8	503.8	647.7	0.8	
25 - 50	38.2	36.5	1.1	1,388.2	1,340.8	1.0	
50 -100	37.1	35.1	1.1	2,547.6	2,534.2	1.1	
100 - 1,000	30.0	29.2	1.0	6,025.1	5,670.1	1.1	
1,000 or more	0.72	0.54	1.3	1,617.6	1,792.6	0.9	
Total	147.5	153.8	1.0	12,170.7	11,895.3	1.0	

Appendix Table 1: Distribution of Returns and Income, Tax Year 2018

Source: Internal Revenue Service Statistics of Income (SOI) and Gale et al (2022b), calculated using Survey of Consumer Finances (SCF). SCF data estimates are calculated using population weights. Data exclude non-filing tax units but include members of the non-primary economic unit who were deemed to be filers.

	1998	2001	2004	2007	2010	2013	2016	2019	Average	Total
	1	1		Sample Siz	ze (Unweight	ed Counts)	1	1		-1
				1	Black Couples	5				
Married	509	584	529	555	920	899	1,037	878	739	5,911
No/One Earner	241	230	226	234	569	504	510	431	368	2,945
Two Earners	268	354	303	321	351	395	527	447	371	2,966
Without Dependents	207	214	233	227	390	346	488	361	308	2,466
With Dependents	302	370	296	328	530	553	549	517	431	3,445
	1			ļ	White Couple:	5	I	1		
Married	10,631	11,356	10,858	11,037	13,557	12,712	13,005	11,915	11,884	95,071
No/One Earner	6,074	6,493	6,085	6,256	7,919	7,387	7,438	6,766	6,802	54,418
Two Earners	4,557	4,863	4,773	4,781	5,638	5,325	5,567	5,149	5,082	40,653
Without Dependents	5,439	5,857	5,718	5,876	7,042	6,697	7,427	7,027	6,385	51,083
With Dependents	5,192	5,499	5,140	5,161	6,515	6,015	5,578	4,888	5,499	43,988
			S	hare of Tax	units that File	e Joint Retur	'n			
Black	25***	26***	22***	27***	24***	24***	22***	24***	24	N/A
White	52	55	52	52	51	50	49	52	52	N/A
Other	53	49	50	53	49	51	55	45	49	N/A
		I	1	Allocation of	Married Cou	iples by Race	9		I	
Black	6	6	6	7	7	8	9	9	7	N/A
White	83	83	80	79	77	76	76	75	78	N/A
Other	11	10	14	15	16	17	16	17	15	N/A

Appendix Table 2: Sample Characteristics

Other111014151617161715N/AAsterisks show ranges of the p-value for differences between the entries of Black couples and the corresponding group of white couples; *** p<0.01, **p<0.05, *p<0.1 For example, 4% of Black couples with AGI below \$50,000 face marriage penalties and that proportion is different, with a p < 0.01, from the 9 percent of white couples with AGI below \$50,000 who face marriage penalties.</td>

Notes: Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty. We define an earner as someone whose sum of wages and business income is greater than zero; couples with negative income (including business losses) are included in the "No-Earner" group. We define a couple as having dependent(s) if they have any dependents age 18 and under.

Appendix Table 3: Allocation Rules for Income and Deductions

Variable	Modeling Assumption
Wages, salaries, and self-employment income	Married scenario:
(excluding qualified business income)	- Total family amount
	"Unmarried" scenario:
	- Allocate family amount between the respondent and spouse in proportion to each's share of the couple's combined wage and salary income
	- If respondent does not report earnings for self and spouse separately, divide family amount equally between respondent and spouse
	- Apply same allocation rules for self-employment income
Interest received (taxable and non-taxable)	Married scenario:
	- Total family amount
	"Unmarried" scenario:
	- Divide equally
Dividend income (qualified dividends only for 2003	Married scenario:
and beyond)	- Total family amount
	"Unmarried" scenario: - Divide equally
Alimony and child support	If there are children under 19 in the family, the full amount is assumed to be child support. Married scenario: - Total family amount
	"Unmarried" scenario:
	- If only one spouse was previously married, that spouse is assumed to receive the full amount
	- Otherwise, divide equally

Schedule C income (active business income)	Married scenario:
	- Total family amount
	"Unmarried" scenario:
	- If only one of the spouses is active in the sole proprietorship, that
	spouse is assumed to receive full amount
	- If both spouses are active in the sole proprietorship and report
	positive business incomes, then full amount is allocated in
	proportion to each spouse's share of the couple's combined
	business income
Net operating loss	Married scenario:
	- Total family amount
	"Unmarried" scenario:
	- Divide equally
Long- and short-term capital gains or losses	Married scenario:
	- Total family amount
	"Unmarried" scenario:
	- Divide equally
Taxable pensions and individual retirement account	Married scenario:
distributions	- Combined amount for respondent and spouse
	"Unmarried" scenario:
	- Respondent and spouse assumed to receive amount reported for
	each individual, respectively
Gross Social Security benefits	Married scenario:
	- Combined amount for respondent and spouse
	"Unmarried" scenario:
	- Respondent and spouse assumed to receive amount reported for
	each individual, respectively

Unemployment compensation received	 Married scenario: Total family amount "Unmarried" scenario: If only one spouse reports being unemployed during the year, the full amount is allocated to that spouse Otherwise, divide evenly
Schedule E income (passive business income)	 Married scenario: Total family amount "Unmarried" scenario: If only one of the spouses is involved in the partnership, S-corporation, limited liability company, or limited liability partnership, that spouse is assumed to receive the full amount If both tax units are owners and have positive business incomes, then allocated in proportion to each's share of the couple's combined business income
Other property income subject to Net Investment Income Tax, including unearned or limited partnership and passive S-corporation profits; rent not eligible for Qualified Business Income deduction; non-qualified dividends; capital gains distributions on form 1040; and other income or loss	 Married scenario: Total family amount "Unmarried" scenario: Allocated to spouse(s) who are (1) self-employed or (2) actively involved in the business. See applicable allocation rules for those income types.
Other taxable income	Married scenario: - Total family amount "Unmarried" scenario: - Divided evenly
Adjustments to income	Married scenario: - Total family amount "Unmarried" scenario: - Allocated to spouse with highest adjusted gross income

Deductions	Married scenario:	
	- Total family amount	
	"Unmarried" scenario:	
	- Allocated to spouse with highest adjusted gross income	

Provision	Law	Modeling Assumption
Filing status	Both must file as single if no dependents or children. Taxpayer may claim head of household filing status if provides over half the costs of maintaining home in which taxpayer resides with children or related dependents.	If there are children, the partner with highest adjusted gross income (AGI) claims head of household status. The other spouse files as single.
Dependent	Generally, the taxpayer who provides more than half the support of a dependent would receive the dependent-related tax benefits. Beginning in 2005, unmarried parents who live together with their children can decide which parent will claim each child as a dependent. If they cannot agree and both claim the same child, then the dependent would be allowed only for the parent with the highest adjusted gross income (AGI).	The partner with highest AGI claims the child.
Child tax credit	See dependent exemption	See dependent exemption
Earned income tax credit (with qualifying children)	If there are children: eligibility is based on which parent has the highest AGI. Beginning in 2002, the AGI tiebreaker rule was relaxed, and unmarried parents can agree which one claims the child if they both live with the child for over half the year. Also beginning in 2002, they can divide their children between them. If they cannot agree and both claim the same child, then the child would be allowed only for the parent with the highest adjusted gross income (AGI).	The partner with highest AGI claims the child.

Appendix Table 4: Allocation Rules for Family-Related Tax Provisions

Earned income tax credit	If there are no children, both may be eligible for	If there are no children, both partners claim the
(without qualifying children)	EITC for those without children.	childless EITC if eligible.
	In 2017, Treasury and the Internal Revenue Service determined that under the code the spouse who does not claim their child is eligible to claim the smaller EITC for workers without children	Otherwise, the partner with the highest AGI claims the EITC for taxpayers with children and the other claims the smaller credit if eligible.
Child and dependent care tax credit, education tax credits, kiddie tax	Omitted from estimates due to insufficient information	on in SCF.

Adjusted Gross Income (\$	А	ll Marri	ed Couples	5	No o		arner Mai iples	rried	Two-Earner Married Couples			
thousands)	Penalty	Bonus	Neither	Total	Penalty	Bonus	Neither	Total	Penalty	Bonus	Neither	Total
				·	All C	Couples	·	·	• •	•		
Less than 50	19	44	37	100	8	47	44	100	58	31	11	100
50 - 100	51	43	6	100	27	71	3	100	72	20	8	100
100 - 200	57	41	2	100	23	76	1	100	75	23	3	100
200 or more	52	48	-	100	34	66	-	100	65	35	1	100
All	43	43	14	100	18	60	22	100	70	25	6	100
					All Blac	k Couple	es					
Less than 50	19	43	38**	100	4***	48***	48	100	62**	29	9*	100
50-100	59***	33***	7***	100	36**	60***	4*	100	73***	17***	10	100
100 - 200	68***	28***	4***	100	29	69**	1	100	83***	12***	5***	100
200 or more	64***	35***	1	100	29	69	2**	100	83***	17***	_***	100
All	46***	36***	18***	100	16***	55***	29***	100	75***	18***	7***	100
	-			·	All Whit	te Coupl	es	·	-			-
Less than 50	18	41	41	100	9	44	47	100	56	32	12	100
50-100	51	44	6	100	28	70	2	100	70	21	8	100
100 - 200	57	41	2	100	24	75	1	100	74	24	2	100
200 or more	51	49	-	100	35	65	-	100	63	36	1	100
All	43	43	14	100	20	59	22	100	69	26	5	100

Appendix Table 5: Prevalence of Marriage Penalties and Bonuses, 2018 Tax Law (%)

Source: Authors' calculations using Survey of Consumer Finances data. Data are derived from a pooled sample of eight waves of the Survey of Consumer Finances (1998 – 2019), and population weights are accordingly divided by eight.

- indicates less than 0.5 percent in absolute terms. Asterisks show ranges of the p-value for differences between the entries of Black couples and the corresponding group of white couples; *** p<0.01, **p<0.05, *p<0.1 For example, 4% of Black couples with AGI below \$50,000 face marriage penalties and that proportion is different, with a p < 0.01, from the 9 percent of white couples with AGI below \$50,000 who face marriage penalties.

Notes: The table shows the percentage of couples – by AGI, race, and number of earners – that face a marriage penalty, a marriage bonus or neither (defined as a bonus or penalty less than \$10 in absolute value). "All Couples" includes all Black couples, all white couples, and all couples of other races. Couples consist of

the SCF respondent and spouse. Both filers and non-filers are included in counts. An earner is someone whose sum of wage and business income is positive. Adjusted gross income (AGI) is computed using 2018 law. Couples with negative labor compensation (comprised of wages, salaries, and business income or losses) are included in the "No-Earner" group. Couples with negative adjusted gross income are not included in the lowest income group but are included in total.

Adjusted	I	All Marrie	d Couples		No or O	ne-Earner	Married	Couples	Two	o-Earner N	Iarried Co	ouples
Gross Income (\$ thousands)	Penalty	Bonus	Neither	Average	Penalty	Bonus	Neither	Average	Penalty	Bonus	Neither	Average
					All	Couples						
Less than 50	-1,181	1,017	0	224	-789	1,056	0	434	-1,391	794	0	-557
50 - 100	-1,303	1,547	0	-8	-1,226	1,719	0	883	-1,327	1,037	0	-748
100 - 200	-2,049	3,012	0	50	-1,669	3,652	0	2,367	-2,111	1,907	0	-1,147
200 or more	-4,880	11,516	0	2,988	-6,899	13,965	0	6,830	-4,103	8,134	0	179
All	-2,064	3,062	0	432	-2,237	3,184	0	1,498	-2,014	2,735	0	-738
	-	·			All Bla	ick Couple	S		-			•
Less than 50	-1,363***	960	0	159**	-935	1,000	0	448***	-1,440*	768	0	-669***
50 - 100	-1,394***	1,402***	0	-358***	-1,151	1,582***	0	540***	-1,466***	1,016	0	-903***
100 - 200	-2,174***	3,147	0	-599***	-1,680	4,023***	0	2,301	-2,241***	1,222***	0	-1,711***
200 or more	-3,664***	10,837	0	1,496***	-3,926***	12,833	0	7,657	-3,614***	6,548*	0	-1,852***
All	-1,804***	1,926***	0	-148***	-1,402***	2,145***	0	946***	-1,888***	1,274***	0	-1,193***
	-	·			All Wh	ite Couple	es		-			•
Less than 50	-1,083	978	0	209	-743	1,012	0	372	-1,324	778	0	-499
50 - 100	-1,241	1,576	0	61	-1,214	1,770	0	907	-1,250	1,022	0	-665
100 - 200	-2,024	2,995	0	90	-1,653	3,649	0	2,331	-2,087	1,922	0	-1,082
200 or more	-4,970	11,609	0	3,139	-6,920	13,983	0	6,643	-4,129	8,307	0	410
All	-2,091	3,304	0	514	-2,329	3,461	0	1,570	-2,017	2,911	0	-645

Appendix Table 6: Average Marriage Penalty (-) or Marriage Bonus (+), 2018 Tax Law (\$)

Source: Authors' calculations using Survey of Consumer Finances data. Data are derived from a pooled sample of eight waves of the Survey of Consumer Finances (1998 - 2019), and population weights are accordingly divided by eight.

Asterisks show ranges of the p-value for differences between the entries of Black couples and the corresponding group of white couples; *** p<0.01, **p<0.05, *p<0.1 For example, 4% of Black couples with AGI below \$50,000 face marriage penalties and that proportion is different, with a p<0.01, from the 9 percent of white couples with AGI below \$50,000 who face marriage penalties.

Notes: The table shows average marriage penalties among couples with penalties and average bonuses among couples with bonuses by AGI, race, and number of earners. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty and are counted as zeroes in

calculating the average figures. "All Couples" includes all Black couples, all white couples, and all couples of other races. Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. An earner is someone whose sum of wage and business income is positive. Adjusted gross income is computed using 2018 law. Couples with negative labor compensation (comprised of wages, salaries, and business income or losses) are included in the "No-Earner" group. Couples with negative adjusted gross income are not included in the lowest income group but are included in total.

Adjusted Gross Income (\$ thousands)	All Married Couples			es	No or One-Earner Married Couples				Two-Earner Married Couples			
	Penalty	Bonus	Neither	Average	Penalty	Bonus	Neither	Average	Penalty	Bonus	Neither	Average
		All Couples										
Less than 50	-3.3	4.0	0.0	1.3	-2.5	3.9	0.0	2.4	-3.8	4.1	0.0	-1.2
50 - 100	-1.9	2.0	0.0	-0.1	-1.7	2.3	0.0	1.1	-1.9	1.3	0.0	-1.1
100 - 200	-1.5	2.3	0.0	0.1	-1.2	2.8	0.0	1.8	-1.5	1.5	0.0	-0.8
200 or more	-1.2	3.0	0.0	0.7	-1.1	3.6	0.0	1.4	-1.2	2.2	0.0	0.0
All	-1.5	2.7	0.0	0.4	-1.2	3.2	0.0	1.5	-1.5	1.9	0.0	-0.6
	All Blac	k Couple	S		-				-			
Less than 50	-3.5	3.8	0.0	1.1	-2.5	3.8	0.0	2.7	-3.6	3.8***	0.0	-1.4
50 - 100	-2.1***	1.8***	0.0	-0.6	-1.6	2.1***	0.0	0.7	-2.2***	1.3***	0.0	-1.4
100 - 200	-1.6	2.3	0.0	-0.4	-1.3	2.9**	0.0	1.7	-1.6***	1.0***	0.0	-1.2
200 or more	-1.2*	3.3*	0.0	0.4	-1.0	4.0**	0.0	1.9	-1.3***	1.9***	0.0	-0.6
All	-1.8	2.6	0.0	-0.2	-1.4	3.1	0.0	1.6	-1.8	1.6	0.0	-1.2
	All Whit	te Couple	es		-				-			
Less than 50	-3.1	3.9	0.0	1.3	-2.3	3.9	0.0	2.1	-3.6	4.0	0.0	-1.0
50-100	-1.8	2.1	0.0	0.0	-1.7	2.3	0.0	1.2	-1.8	1.3	0.0	-1.0
100 - 200	-1.4	2.3	0.0	0.1	-1.2	2.8	0.0	1.8	-1.5	1.5	0.0	-0.8
200 or more	-1.2	2.9	0.0	0.7	-1.1	3.5	0.0	1.3	-1.2	2.2	0.0	0.1
All	-1.4	2.7	0.0	0.4	-1.2	3.1	0.0	1.4	-1.5	1.9	0.0	-0.5

Appendix Table 7: Average Marriage Penalty Rate (-) or Bonus Rate (+), 2018 Tax Law (percent of AGI)

Source: Authors' calculations using Survey of Consumer Finances data. Data are derived from a pooled sample of eight waves of the Survey of Consumer Finances (1998 - 2019), and population weights are accordingly divided by eight.

Asterisks show ranges of the p-value for differences between the entries of Black couples and the corresponding group of white couples; *** p<0.01, **p<0.05, *p<0.1 For example, 4% of Black couples with AGI below \$50,000 face marriage penalties and that proportion is different, with a p<0.01, from the 9 percent of white couples with AGI below \$50,000 who face marriage penalties.

Notes: The table shows average marriage penalty rates (i.e., the penalty as a share of AGI) among couples with penalties and average bonus rates among couples with bonuses by AGI, race, and number of earners. Couples with bonuses or penalties of less than \$10 (absolute value) are treated as having neither a bonus nor penalty and are counted as zeroes in calculating the average figures. "All Couples" includes all Black couples, all white couples, and all couples of other races. Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. An earner is someone whose sum of wage and business income is positive. Adjusted gross income is computed using 2018 law. Couples with negative labor compensation (comprised of wages, salaries, and business income or losses) are included in the "No-Earner" group. Couples with negative adjusted gross income are not included in the lowest income group but are included in total.

Adjusted Gross	A	ll Marri	ed Couples		Coup	les with 1	No Depen	dents	Co	ouples wi	th Depen	dents
Income (\$ thousands)	Penalty	Bonus	Neither	Total	Penalty	Bonus	Neither	Total	Penalty	Bonus	Neither	Total
					All C	Couples						
Less than 50	19	44	37	100	12	38	50	100	29	53	18	100
50 - 100	51	43	6	100	36	52	12	100	64	36	1	100
100 - 200	57	41	2	100	44	52	5	100	67	33	-	100
200 or more	52	48	-	100	52	47	1	100	51	49	-	100
All	43	43	14	100	30	46	24	100	55	40	5	100
					Black	Couples						
Less than 50	19	43	38**	100	8****	40	52	100	31	47*	22**	100
50 - 100	59***	33***	7***	100	33*	49*	18***	100	75***	24***	1	100
100 - 200	68***	28***	4***	100	52***	37***	11***	100	77***	23***	-***	100
200 or more	64***	35***	1	100	52	48	-1***	100	71***	28***	1**	100
All	46***	36***	18***	100	26***	42***	32***	100	62***	31***	7***	100
					White	Couples						
Less than 50	18	41	41	100	12	37	51	100	31	50	19	100
50 - 100	51	44	6	100	37	52	11	100	63	36	1	100
100 - 200	57	41	2	100	43	53	4	100	68	32	-	100
200 or more	51	49	-0	100	53	46	1	100	49	51	-	100
All	43	43	14	100	31	46	23	100	57	40	4	100

Appendix Table 8: Prevalence of Marriage Penalties and Bonuses by Dependent Presence, 2018 Tax Law (%)

Source: Authors' calculations using Survey of Consumer Finances data. Data are derived from a pooled sample of eight waves of the Survey of Consumer Finances (1998 – 2019), and population weights are accordingly divided by eight.

- indicates less than 0.5 percent in absolute terms. Asterisks show ranges of the p-value for differences between the entries of Black couples and the corresponding group of white couples; *** p<0.01, **p<0.05, *p<0.1 For example, 4% of Black couples with AGI below \$50,000 face marriage penalties and that proportion is different, with a p < 0.01, from the 9 percent of white couples with AGI below \$50,000 who face marriage penalties.

Notes: The table shows the percentage of couples – by AGI, race, and number of earners – that face a marriage penalty, a marriage bonus or neither (defined as a bonus or penalty less than \$10 in absolute value). "All Couples" includes all Black couples, all white couples, and all couples of other races. Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. An earner is someone whose sum of wage and business income is positive. Adjusted gross income is computed using 2018 law. Couples with negative labor compensation (comprised of wages, salaries, and business income or losses) are included in the "No-Earner" group. Couples with negative adjusted gross income are not included in the lowest income group but are included in total. Couples who are noted as having dependents have dependents aged 18 and under.

Adjusted		All Marri	ed Couple	S	Coup	les with N	o Depend	ents	C	ouples wit	h Depend	ents
Gross Income (\$ thousands)	Penalty	Bonus	Neither	Total	Penalty	Bonus	Neither	Total	Penalty	Bonus	Neither	Total
					All Co	ouples						
Less than 50	-1181	1017	0	224	-424	843	0	270	-1640	1205	0	155
50 - 100	-1303	1547	0	-8	-1017	1864	0	601	-1429	1187	0	-484
100 - 200	-2049	3012	0	50	-1606	3057	0	881	-2259	2961	0	-557
200 or more	-4880	11516	0	2988	-4719	10874	0	2611	-5012	12007	0	3289
All	-2064	3062	0	432	-1820	2850	0	755	-2190	3288	0	128
	-				Black (Couples						
Less than 50	-1363***	960	0	159**	-295***	795	0	294*	-1657	1110**	0	14*
50 - 100	-1394***	1402***	0	-358***	-943*	1633***	0	485*	-1516***	1113	0	-877***
100 - 200	-2174***	3147	0	-599***	-1747***	3502**	0	408***	-2335**	2816	0	-1166***
200 or more	-3664***	10837	0	1496***	-3134***	9367*	0	2819	-3900***	12326	0	706***
All	-1804***	1926	0	-148***	-1348***	1937***	0	464***	-1948***	1916***	0	-605***
					White (Couples						
Less than 50	-1083	978	0	209	-443	841	0	260	-1613	1198	0	97
50 - 100	-1241	1576	0	61	-1027	1867	0	599	-1353	1198	0	-422
100 - 200	-2024	2995	0	90	-1585	3023	0	921	-2244	2959	0	-568
200 or more	-4970	11609	0	3139	-4842	10935	0	2530	-5090	12147	0	3670
All	-2091	3304	0	514	-1890	2958	0	777	-2211	3745	0	226

Appendix Table 9: Average Marriage Penalty (-) or Marriage Bonus (+) by Dependent Presence, 2018 Tax Law (\$)

Source: Authors' calculations using Survey of Consumer Finances data. Data are derived from a pooled sample of eight waves of the Survey of Consumer Finances (1998 – 2019), and population weights are accordingly divided by eight.

- indicates less than 0.5 percent in absolute terms. Asterisks show ranges of the p-value for differences between the entries of Black couples and the corresponding group of white couples; *** p<0.01, **p<0.05, *p<0.1 For example, 4% of Black couples with AGI below \$50,000 face marriage penalties and that proportion is different, with a p < 0.01, from the 9 percent of white couples with AGI below \$50,000 who face marriage penalties.

Notes: The table shows the percentage of couples – by AGI, race, and number of earners – that face a marriage penalty, a marriage bonus or neither (defined as a bonus or penalty less than \$10 in absolute value). "All Couples" includes all Black couples, all white couples, and all couples of other races. Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. An earner is someone whose sum of wage and business income is positive.

Adjusted gross income is computed using 2018 law. Couples with negative labor compensation (comprised of wages, salaries, and business income or losses) are included in the "No-Earner" group. Couples with negative adjusted gross income are not included in the lowest income group but are included in total. Couples who are noted as having dependents have dependents aged 18 and under.

Adjusted Gross		All C	ouples		Coup	les with]	No Depen	dents	Co	ouples wi	th Depen	dents
Income (\$ thousands)	Penalty	Bonus	Neither	Total	Penalty	Bonus	Neither	Total	Penalty	Bonus	Neither	Total
					All C	ouples						
Less than 50	-3.3	4.0	0.0	1.3	-1.5	2.9	0.0	1.4	-4.4	5.1	0.0	1.2
50 - 100	-1.9	2.0	0.0	-0.1	-1.4	2.5	0.0	0.8	-2.1	1.5	0.0	-0.8
100 - 200	-1.5	2.3	0.0	0.1	-1.1	2.4	0.0	0.7	-1.6	2.2	0.0	-0.4
200 or more	-1.2	3.0	0.0	0.7	-1.0	2.7	0.0	0.6	-1.3	3.1	0.0	0.8
All	-1.5	2.7	0.0	0.4	-1.1	2.6	0.0	0.7	-1.7	2.8	0.0	0.2
					Black	Couples						
Less than 50	-3.5***	3.8	0.0	1.1	-1.1	2.8	0.0	1.8	-4.1	4.6***	0.0	0.6
50 - 100	-2.1***	1.8***	0.0	-0.6	-1.3***	2.1***	0.0	0.6	-2.3***	1.5	0.0	-1.4
100 - 200	-1.6***	2.3	0.0	-0.4	-1.3	2.5	0.0	0.3	-1.7***	2.1	0.0	-0.8
200 or more	-1.2**	3.3*	0.0	0.4	-1.0	3.1*	0.0	0.9	-1.3	3.5	0.0	0.1
All	-1.8	2.6	0.0	-0.2	-1.2	2.5	0.0	0.7	-2.0	2.7	0.0	-0.7
					White	Couples						
Less than 50	-3.1	3.9	0.0	1.3	-1.5	2.9	0.0	1.3	-4.4	5.5	0.0	1.2
50 - 100	-1.8	2.1	0.0	0.0	-1.4	2.5	0.0	0.8	-2.0	1.5	0.0	-0.7
100 - 200	-1.4	2.3	0.0	0.1	-1.1	2.3	0.0	0.7	-1.6	2.2	0.0	-0.4
200 or more	-1.2	2.9	0.0	0.7	-1.0	2.7	0.0	0.6	-1.3	3.1	0.0	0.9
All	-1.4	2.7	0.0	0.4	-1.1	2.6	0.0	0.7	-1.6	2.8	0.0	0.2

Appendix Table 10: Average Marriage Penalty Rate (-) or Bonus Rate (+) by Dependent Presence, 2018 Tax Law (percent of AGI)

Source: Authors' calculations using Survey of Consumer Finances data. Data are derived from a pooled sample of eight waves of the Survey of Consumer Finances (1998 – 2019), and population weights are accordingly divided by eight.

Asterisks show ranges of the p-value for differences between the entries of Black couples and the corresponding group of white couples; *** p<0.01, **p<0.05, *p<0.1 For example, 4% of Black couples with AGI below \$50,000 face marriage penalties and that proportion is different, with a p < 0.01, from the 9 percent of white couples with AGI below \$50,000 who face marriage penalties.

Notes: The table shows the percentage of couples – by AGI, race, and number of earners – that face a marriage penalty, a marriage bonus or neither (defined as a bonus or penalty less than \$10 in absolute value). "All Couples" includes all Black couples, all white couples, and all couples of other races. Couples consist of the SCF respondent and spouse. Both filers and non-filers are included in counts. An earner is someone whose sum of wage and business income is positive. Adjusted gross income is computed using 2018 law. Couples with negative labor compensation (comprised of wages, salaries, and business income or losses) are included in the "No-Earner" group. Couples with negative adjusted gross income are not included in the lowest income group but are included in total. Couples who are noted as having dependents have dependents aged 18 and under.

Provisions Intended to Reduce Marriage Penalties	Other Key Provisions with Potential Effects on Marriage Bonuses and Penalties
Fully Phased-in Provisions in	Economic Growth and Tax Relief Reconciliation Act of 2001 (EGTRRA) and the Jobs and Growth Tax Relief Reconciliation Act of 2003 (JGTRRA) ²⁶
Prior law: The ratio of the width of the 15 percent rate bracket for	Prior law: Individual income tax rates ranged from 15 percent to 39.6 percent.
married couples filing jointly (joint filers) to the width of that bracket	Policy: Reduced income tax rates to between 10 percent and 35 percent.
for single filers was set at 1.67:1.	Effect: For most taxpayers, lowering the rates reduced marriage penalties and bonuses. However, no changes were made to the tax rates for the alternative minimum tax (AMT), and the AMT thresholds for married filers
New law: Increased that ratio to 2:1 for both the 15 percent rate bracket and the new 10 percent rate bracket.	remained at less than twice the levels for unmarried filers. The interaction between the reduced regular income tax rates and the AMT increased the probability of becoming subject to the AMT and incurring marriage penalties— especially for couples with high income, large families, or sizable state and local taxes.
Effect: A ratio of 2:1 is marriage neutral. A ratio of less than 2 causes marriage penalties.	
Prior law: The ratio of the standard deduction for joint filers to the amount for single filers was set at	Prior law: The ratio of the standard deduction for joint filers to the amount for head-of-household filers was set at 1.14:1.
1.67:1.	New law: No change
New law : Increased that ratio to 2:1.	Effect: The ratio of the standard deduction for joint filers to that for heads of households rose to 1.36:1 due to the statutory increase in the former.
Effect: A ratio of 2:1 is marriage neutral.	
Prior law: The earned income tax credit (EITC) began to phase out at the same income level for both	Prior law: If an unmarried couple had children and lived together, only the parent with the highest adjusted gross income (AGI) was eligible for the EITC.
unmarried and joint filers—a ratio of 1:1.	New law: Unmarried cohabitating parents could choose which one claimed their children for the EITC. In addition, they could split their children (e.g., in a two-child family, each could claim one of the children).

Appendix Table 11: Legislation with Direct and Indirect Effects on Marriage Bonuses and Penalties, 2001-2017

²⁶ Most provisions were phased in, with full implementation by 2010. Those provisions, however, were scheduled to expire at the end of 2010. With the exception of a few provisions, they were extended permanently by the American Taxpayer Relief Act of 2012.

New law: Increased the starting point by \$3,000 for joint filers relative to unmarried filer—a ratio of 1.19:1 (for couples with children)	Effect: Gave unmarried couples more flexibility to maximize the household's total EITC relative to married couples, thus increasing marriage penalties or reducing marriage bonuses.
Effect: Reduced but did not eliminate marriage penalties in EITC.	
	Prior law: Personal exemptions began to phase out when AGI exceeded \$128,950 for singles, \$161,150 for heads of households, and \$193,400 for joint filers.
	New law: Phaseout repealed.
	Effect: Because the threshold at which exemptions had begun to phase out for joint filers was less than half the level as for unmarried filers, repealing the provision reduced marriage penalties.
	Prior law: The sum of itemized deductions was limited if AGI exceeded \$128,950.
	New law: Limitation on itemized deductions repealed.
	Effect: Because the threshold at which itemized deductions had been limited for married couples was the same as for unmarried filers, repealing the provision reduced marriage penalties.
	Prior law: The maximum amount of the child tax credit was \$500, and the credit began to phase out when AGI exceeded \$75,000 for unmarried filers and \$110,000 for joint filers (a ratio of 1.46:1). The credit was refundable only for taxpayers with three or more children who met certain other criteria.
	New law: The credit amount was doubled. The credit became partially refundable, phasing in at 15 percent of earned income in excess of \$10,000 (indexed for inflation).
	Effect: Because the phaseout thresholds were not changed, the larger amount of the credit resulted in higher marriage penalties for couples affected by the phaseout. The phased-in refundable tax credit could result in marriage bonuses—if, for example, a childless worker married a nonworking parent.
	conomic Stabilization Act of 2008 and American Recovery and Relief Act of 2009
Prior law: The EITC phaseout began at a higher level for joint	Prior law: The EITC amount increased with the presence and number of children, up to two or more children.

filers than for unmarried filers: ratio of 1.19:1.	New law: Added a fourth tier to the EITC, so that the EITC was now higher for families with three or more children than for smaller families.
New law: Increased the starting point to \$5,000 more for married couples relative to unmarried filer—a ratio of 1.30:1.	Effect: The provision increased marriage bonuses and reduced marriage penalties for some couples with larger families (e.g., a worker with two children could marry a nonworking parent with one child and receive a larger EITC).
Effect: Reduced but did not eliminate marriage penalties in EITC.	
	Prior law: The refundable portion of the child tax credit was phased in, beginning at \$10,000 (indexed for inflation).
	New law: Temporarily lowered the earnings threshold for eligibility for the refundable portion of the child tax credit to \$8,500 in 2008 and \$3,000 in 2009.
	Effect: Increased marriage bonuses for very low-income income working families (e.g., a childless worker married to a nonworking parent).
	Affordable Care Act of 2010
	New law: Established premium tax credits for individuals and families with income between 100 percent and 400 percent of the poverty line, with the amount of the subsidy declining over that range.
	Effect: Because the official poverty line does not increase linearly with family size, the premium tax credits can result in marriage penalties (e.g., in combination, two unmarried parents could be eligible for larger credits than if they married).
	New law : Created the net investment income tax, starting at \$200,000 for unmarried filers and \$250,000 for married couples filing jointly—a ratio of 1.25:1.
	Effect: Some couples incur a marriage penalty because the threshold for joint filers is less than twice the amount as for unmarried filers.
	New law: Created the Additional Medicare Tax, starting at \$200,000 of wages for unmarried filers and \$250,000 for married couples filing jointly—a ratio of 1.25:1.
	Effect: Some couples incur a marriage penalty because the threshold for joint filers is less than twice the amount as for unmarried filers.

	American Taxpayer Relief Act of 2012
	Prior law: EGTRRA's tax rate reductions were scheduled to expire at end of 2012.
	New law: EGTRRA's tax rate reductions were extended—except for the reduction in the top rate bracket.
	Effect: Restored the top income tax rate of 39.1 percent. Because the top bracket for joint filers begins at
	\$450,000—less than twice the level as for single filers (\$400,000)—marriage penalties increased for high-income taxpayers relative to the prior year.
	Prior law: The phaseout of the personal exemption was scheduled to be restored.
	New law: The phaseout of the personal exemption was restored.
	Effect: Because the phaseout began at \$300,000 for joint filers—less than twice the threshold (\$250,000) for single filers—marriage penalties increased for higher-income taxpayers relative to prior year.
	Prior law: Limitation on total itemized deductions was scheduled to be restored in 2013.
	New law: The limitation on itemized deductions was restored.
	Effect: Because the limitation began at \$300,000 for joint filers—less than twice the threshold (\$250,000) for single filers—marriage penalties increased for higher-income taxpayers relative to prior year.
	Prior law: Child tax credit parameters scheduled to be restored to pre-EGTRRA levels.
	New law: Retained \$1,000 maximum credit and partial refundability and lower threshold for refundability.
	Effect: Maintained increased marriage bonuses for very low-income childless workers who married nonworking parent.
	Tax Cuts and Jobs Act of 2017 (TCJA) ²⁷
Prior law: Rate brackets for joint filers were less than twice the width	Prior law: Individual income tax rates ranged from 10 percent to 39.6 percent.
of single filers' brackets, other than for the 10 percent and 15 percent	New law: Individual income tax rates ranged from 10 percent to 37 percent.
brackets.	Effect: Reduced individual income tax rates, thus reducing marriage penalties. Because the AMT thresholds were substantially increased, the interaction between the lower rates on ordinary income and the AMT was markedly reduced relative to the impact in EGTRRA.

²⁷ The individual income tax provisions in TCJA are scheduled to expire at the end of 2025.

Prior law: Personal and dependent exemption began to phase out for married couples when income exceeded a
threshold that was less than twice the level as for single filers.
New law: Eliminated the personal and dependent exemption.
Effect: Eliminating the exemption reduced marriage penalties by also effectively eliminating the phaseout.
Prior law: The limitation on itemized deductions reduced a taxpayer's itemized deductions above a certain
threshold of income. This threshold for married couples was less than twice the level as for single filers.
New law: Eliminated the limitation on itemized deduction
Effect: Elimination of limitation reduced marriage penalties.
Prior law: No limit on amount of state and local taxes that can be itemized (if taxpayer has sufficient amount of
deductions to make itemizing more valuable than standard deduction).
New law: Imposed a \$10,000 cap on deductible state and local taxes. The cap is the same amount for unmarried and married filers.
Effect: Flat limit, without regard to filing status, increased marriage penalties.
Prior law: The maximum of the child tax credit was \$1,000 per qualifying child, phasing out beginning at \$75,000 for unmarried filers and \$110,000 for joint filers. Eligibility for the refundable portion began at \$3,000 of earned income.
New law: Increased the maximum amount of the nonrefundable portion of the child tax credit to \$2,000 per qualifying child, phasing out beginning at \$200,000 for unmarried filers and \$400,000 for joint filers. The refundable portion of the credit was capped at \$1,400 per child, and the income threshold for eligibility was lowered to \$2,500.

Effect: Setting the phaseout threshold for married couples to be twice that of unmarried couples eliminated
marriage penalties attributable to that region. The effects in the phase-in range are less clear: the higher amount of
the credit and the lower threshold for refundability increases bonuses for some couples, but the cap on the
refundable portion might offset that impact.



Appendix Figure 1: Aggregate Tax Liability After Credits, Statistics of Income and Survey of Consumer Finances

Source: Gale et al. (2022b)